

**Delft University of Technology** 

# A General Framework to Forecast the Adoption of Novel Products

# A Case of Autonomous Vehicles

Dubey, Subodh; Sharma, Ishant; Mishra, Sabyasachee; Cats, Oded; Bansal, Prateek

DOI 10.1016/j.trb.2022.09.009

Publication date 2022 **Document Version** Final published version

Published in Transportation Research Part B: Methodological

## Citation (APA)

Dubey, S., Sharma, I., Mishra, S., Cats, O., & Bansal, P. (2022). A General Framework to Forecast the Adoption of Novel Products: A Case of Autonomous Vehicles. *Transportation Research Part B:* Methodological, 165, 63-95. https://doi.org/10.1016/j.trb.2022.09.009

## Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright

Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

Contents lists available at ScienceDirect







journal homepage: www.elsevier.com/locate/trb

# A General Framework to Forecast the Adoption of Novel Products: A Case of Autonomous Vehicles



Subodh Dubey <sup>a,1</sup>, Ishant Sharma <sup>b,1</sup>, Sabyasachee Mishra <sup>b</sup>, Oded Cats <sup>a</sup>, Prateek Bansal <sup>c,\*</sup>

<sup>a</sup> Department of Transport and Planning, Delft University of Technology, Netherlands

<sup>b</sup> Department of Civil Engineering, The University of Memphis, Tennessee, United States

<sup>c</sup> Department of Civil and Environmental Engineering, National University of Singapore, Singapore

#### ARTICLE INFO

Keywords: Word of mouth Risk aversion Novel technology adoption Social network Autonomous vehicles

#### ABSTRACT

Due to the unavailability of prototypes, the early adopters of novel products actively seek information from multiple sources (e.g., media and social networks) to minimize the potential risk. The existing behavior models not only fail to capture the information propagation within the individual's social network, but also they do not incorporate the impact of such word-of-mouth (WOM) dissemination on the consumer's risk preferences. Moreover, even cutting-edge forecasting models rely on crude/synthetic consumer behavior models. We propose a general framework to forecast the adoption of novel products by developing a new consumer behavior model and integrating it into a population-level agent-based model. Specifically, we extend the hybrid choice model to estimate consumer behavior, which incorporates social network effects and interplay between WOM and risk aversion. The calibrated consumer behavior model and synthetic population are passed through the agent-based model for forecasting the product market share. We apply the proposed framework to forecast the adoption of autonomous vehicles (AVs) in Nashville, USA. The consumer behavior model is calibrated with a stated preference survey data of 1,495 Nashville residents. The output of the agent-based model provides the effect of the purchase price, post-purchase satisfaction, and safety measures/regulations on the forecasted AV market share. With an annual AV price reduction of 5% at the initial purchase price of \$60,000 and 90% of satisfied adopters, AVs are forecasted to attain around 80% market share in thirty-one years. These findings are crucial for policymakers to develop infrastructure plans and manufacturers to conduct an after-sales cost-benefit analysis.

#### 1. Introduction

Capturing consumers' intention to purchase or adopt a novel product is vital for multiple disciplines such as transportation, marketing, sales, technology, economics, finance, human-machine interaction, and social behavior, among others. We contribute to this interdisciplinary literature by providing a general framework to elicit consumers' preferences and forecasting their adoption of

\* Corresponding Author: Dr Prateek Bansal, National University of Singapore, Singapore

*E-mail addresses*: s.k.dubey@tudelft.nl (S. Dubey), isharma@memphis.edu (I. Sharma), smishra3@memphis.edu (S. Mishra), o.cats@tudelft.nl (O. Cats), prateekb@nus.edu.sg (P. Bansal).

<sup>1</sup> Authors contributed equally to this work.

https://doi.org/10.1016/j.trb.2022.09.009

Received 7 September 2021; Received in revised form 27 September 2022; Accepted 30 September 2022

Available online 12 October 2022

<sup>0191-2615/© 2022</sup> The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

"really new products" (i.e., innovative products with entirely new or different attributes from any existing products) (Gregan-Paxton and John, 1997). Autonomous vehicle (AV) – a fully-automated self-driving privately-owned vehicle – is a case in point, which falls under this product category with recent innovations in technology-assisted motorized driving. Understanding consumer preferences and forecasting adoption rates of AVs are crucial for policymakers to devise a plan to meet infrastructure needs and make regulatory decisions to manage a mixed fleet of AVs and conventional vehicles (CVs). At the same time, quantifying the effect of the purchase price and marketing strategies on AV adoption rate is equally critical for manufacturers to conduct an after-sales cost-benefit analysis.

## 2. Background and Motivation

Potential consumers can state their intentions to adopt an existing product based on the attributes of interest, by means of trial periods and test drives. In contrast, capturing consumer's intention to purchase a futuristic product is challenging due to the unavailability of accessible prototypes for first-hand experience. Therefore, early adopters actively search for information about the anticipated attributes, benefits, and barriers associated with novel products to minimize associated risks and maximize post-purchase satisfaction (Dholakia, 2001; Dowling and Staelin, 1994; Manning et al., 1995).

With recent technological advancements in smartphones and ubiquitous internet connectivity, potential consumers are increasingly exchanging their opinions and recommendations about innovations through electronic media, social media, blogs, and peer-topeer communication, among other communication channels and informational sources (Gupta and Harris, 2010; Huete-Alcocer, 2017). The information obtained from such channels is commonly referred to as *word-of-mouth* (WOM). In this era of a digital revolution, the influence of the product-related information through WOM on consumer preferences has become substantial enough to be carefully accounted for in econometric models (Huete-Alcocer, 2017). Due to the inability to experience the product, WOM plays an even more vital role in alleviating or increasing risks associated with the adoption of novel product (Hirunyawipada and Paswan, 2006; Hussain et al., 2018, 2017; Krishnamurthy, 2001; Manning et al., 1995; Tan, 1999). Such differences in product information transmission and its effect on the consumers' risk perception call for specific consumer behavior models for novel products.

We envision that an ideal econometric model to elicit the consumers' preferences for novel products should have five components: consumer's risk preferences, WOM through offline social networks and online channels, the interplay between consumer's risk preferences and WOM, adoption level of the product in social network/city, and the influence of key product attributes such as purchase price on consumer's preferences. Followed by generating a synthetic population, such a consumer preference model can be integrated into an agent-based model to forecast the adoption of the novel product under different scenarios (e.g., purchase price reduction or changes in risk preferences due to technological improvements such as reduction in AV crash rate).

#### 2.1. Research Gap

The literature on capturing the impact of WOM on consumer preferences is prolific (see Table A.1 in supplementary material for a summary of 40+ such studies). Specific to the transportation sector, structural equation models (Baber et al., 2016; Chatura, 2015; Chen et al., 2016; Hussain et al., 2018, 2017; Iyer and Griffin, 2020; Jalilvand and Samiei, 2012; Kwon et al., 2020; Martin and Lueg, 2013; Thøgersen and Ebsen, 2019; Torlak et al., 2014; Tsiotsou and Alexandris, 2009), discrete choice models (He et al., 2014; Helveston et al., 2015; Jansson et al., 2017), agent-based models (Kieckhäfer et al., 2017), exploratory factor analysis (Ozaki and Sevastyanova, 2011), regression analysis (Barth et al., 2016; Bhandari and Rodgers, 2018; Cheung et al., 2009; Du et al., 2018; Jung and Seock, 2017; Marchand et al., 2017; Moons and De Pelsmacker, 2012), text mining (Ma et al., 2019), theory of reasoned action (Alzahrani et al., 2019; Prendergast et al., 2010), and Bass model (Hong et al., 2020; Huang et al., 2014; Mahajan et al., 1984) have been used to model WOM and social network effects in the adoption of green or electric vehicles (EVs). Specific to AVs, Ghasri and Vij (2021) explored the influence of WOM on the consumer preferences using discrete choice. Most past studies could only quantify consumers' preferences to adopt EVs or AVs, but failed to translate the estimated preferences into a forecasting model. For instance, the results of structural equation models on *consumers' intention to purchase* are not adequate for forecasting the adoption of novel products. Only He et al. (2014) and Kieckhäfer et al. (2017) extended the analysis to incorporate the effect of WOM in forecasting the adoption of EVs, and Talebian and Mishra (2018) did the same for AVs. However, these studies lack the underlying consumer behaviour model.

Similarly, only a handful of previous studies used structural equation models (Chikaraishi et al., 2020; Ha, 2002; Hirunyawipada and Paswan, 2006), the technology acceptance model (Zhang et al., 2019), or discrete choice models (Bansal et al., 2021b; Wang and Zhao, 2019) to understand consumers' risk preferences in the adoption of EVs and AVs. However, none of these studies forecasted the market penetration of new technologies. A similar pattern was observed in modeling risk preferences for other novel products such as farming techniques (Barham et al., 2014; Brick and Visser, 2015). To substantiate our claim, we summarize the related literature in Table A.1 in supplementary material. There are also many studies in the literature, which completely ignored WOM and risk preferences, and relying exclusively on product attributes (such as purchase price) in agent-based models to forecast the adoption of novel products such as EVs and AVs (Bansal and Kockelman, 2017; Musti and Kockelman, 2011).

In summary, numerous studies modeled WOM and risk preferences in eliciting consumers' inclination towards novel products, but they have three main shortcomings. First, previous studies fail to explicitly incorporate the social network effect. A handful of studies have modeled the effect of information from internal or external sources on consumer preferences (Ghasri and Vij, 2021; Simpson et al., 2022; Talebian and Mishra, 2022; Sharma and Mishra, 2022a, 2022b, 2020)) but have failed to model *information propagation* within the social network. Therefore, these consumer behavior models cannot be used to forecast the adoption of novel products. Second, none of the previous studies have gone beyond consumer behavior analysis and forecasted the adoption of the novel product,

forecasting models rely on simplistic (mostly synthetic, i.e., not calibrated with the contextual data) consumer behavior models.

#### 2.2. Contributions

We propose a comprehensive framework to forecast the adoption of the novel product where a well-calibrated new consumer behavior model is integrated into a population-based agent-based model. Specifically, the consumer behavior model follows the specification of the integrated choice and latent variable (ICLV) model (also known as the hybrid choice model), where WOM and risk aversion are considered as latent variables. The proposed consumer behavior, interdependent ICLV henceforth, builds upon Bhat et al. (2016b)'s spatial ICLV framework by incorporating cross-loading of latent variables and panel effects in the discrete choice component. In this specification, autoregressive structure in the latent construct of ICLV captures social network effects, and cross-loading WOM on risk aversion accounts for their interaction effects. The indirect utility of the choice model captures the effect of the purchase price and adoption rates of the product within the social network and the city. We derive the maximum likelihood estimator of the interdependent ICLV model and calibrate it with stated preference data. The calibrated consumer behavior model, synthetic population, and individual-level social network are passed through an agent-based simulation model to predict individual's preferences to buy the novel product in each time step. The individual-level preferences are aggregated to find the adoption rate in each time step. We also highlight the implications of neglecting the information propagation effect in the agent-based simulation. The proposed framework is general enough to be applied for forecasting the adoption of any novel product. However, to make the discussion contextual, we demonstrate its capabilities in forecasting the adoption of AVs.

The loading matrix would be a lower-triangular matrix to ensure one-way cross-loading/structural effect. The remainder of the paper is organized as follows: Section 2 discusses the design of stated choice experiments to collect the preferences of Nashville (Tennessee, USA) residents for AV adoption. The details of data collection and summary statistics are also presented in this section. Section 3 provides specific details of the considered aspects of WOM and risk preferences, followed by a contextual and mathematical representation of the interdependent ICLV model. Section 4 summarizes the results of the ICLV model. Section 5 details the synthetic population generation, the agent-based simulation framework, and scenario-based analysis of the forecasted adoption rates of AVs in Nashville. Conclusions and avenues of future research are discussed in the final section.

## 3. Survey Design and Data Collection

## 3.1. Discrete Choice Experiment Design

We designed and conducted a stated preference survey with a discrete choice experiment (DCE). In the DCE, respondents were asked to choose between a CV and an AV during their next car purchase. They needed to make choices based on the purchase prices of both cars and the adoption of AVs in their social network and city. Before the experiment, respondents were informed about the differences between AVs and CVs using infographics. To ensure that respondents do not find automation very futuristic, we also mentioned that Google's AV has driven more than 20 million miles on public roads. We communicated that Level 4 AVs are slightly more expensive than CVs because they need additional accessories to operate without a human driver, but both cars are equivalent in all other attributes (e.g., engine power, fuel economy, body type, and aesthetics). Attribute levels of both alternatives in the DCE are presented in Table 1. The purchase price of AV was pivoted on the purchase price of the CV, which was asked in a question preceding the DCE.

Four attribute levels for the purchase price and three attribute levels for each social network and city level AV adoption lead to a total of thirty-six choice scenarios. We used the full factorial design and presented each respondent with three randomly selected scenarios out of thirty-six choice scenarios. An example of the choice scenario presented to respondents is shown in Fig. 1.

## 3.2. WOM Experiments

Apart from the DCE, the survey also had two experiments to capture the type and magnitude of the WOM transmitted by respondents based on the source of information and AV attributes. In both experiments, we used four information sources: friend, car dealer, colleague, and media. All the attribute levels of both experiments are tabulated in Table 1.

In experiment 1, we provided information about vehicle crashes and fatality rates for CVs and AVs, and the percentage of AV crashes with no clarity about who is responsible for the crash. Following Blanco et al. (2016), we used 1,090 and 415 vehicle crashes per 100 million miles as crash rates for CVs and AVs, respectively. Assuming a potential reduction of 30-50% in the number of AV crashes, we considered 415, 290, and 207 as the three levels of the AV crash rate. Whereas we used 1.13 fatalities per 100 million miles for CVs (IIHS, 2020), it was considered to be zero for AVs because Google's AV did not report any fatality in the year 2018 and 2019 (Waymo, 2020). We also assumed that making anyone accountable or responsible for the crash might be challenging in 10-30% of AV crashes. Therefore, we considered 10% and 30% as two levels for this attribute. Experiment 2 is similar to experiment 1, but crash rates and fatalities were replaced with reductions in travel time and CO<sub>2</sub> emissions due to automation. The past findings suggest that AVs are expected to reduce travel time and CO<sub>2</sub> emissions by 37% and 30%, respectively, at 50% market penetration (Olia et al., 2016). Based on this information, we used attribute levels of 20% and 40% for travel time reduction and 30% and 50% for CO<sub>2</sub> emission reduction.

The full factorial designs of experiments 1 and 2 have  $(4 \times 3 \times 2)$  24 and  $(4 \times 2 \times 2 \times 2)$  32 choice scenarios. One randomly selected scenario was presented to each respondent for one of the two experiments. Readers will note that both experiments ensure a trade-off between the benefits of automation (crash/fatality reduction in experiment 1, and travel time and emissions reduction in experiment

Attribute	Conventional car	Autonomous car
Purchase price	42500	63750
Percentage of people in your <b>social</b> <b>network</b> who adopted/bought autonomous cars		60%
Percentage of people in <b>your city</b> who adopted/bought autonomous cars		30%

Would you be **willing to pay extra and buy** the autonomous vehicle mentioned above?

Yes, I would buy this autonomous vehicle during the next purchase.

No, I would stick to a conventional vehicle during the next purchase.

Fig. 1. An example of a choice situation presented to respondents.

2) and the associated risks (AV crashes with no clarity on responsibility). Considering these trade-offs, the respondent was asked to rate three statements, each corresponding to positive, neutral, and negative WOM, on a five-point Likert scale (from strongly disagree to strongly agree) in each experiment. Specifically, based on the presented information in the experiment, the respondents were asked how likely they would positively, neutrally, or negatively communicate their opinion about AV adoption to their close social ties. The narratives of both experiments 1 and 2, along with the WOM statements, are shown in example scenarios presented in Fig. 2 and Fig. 3, respectively.

## 3.3. Data Collection

The web-based stated preference survey was hosted on Qualtrics and was disseminated among Nashville residents between August and November 2020. Survey participants were recruited with the help of an online market research firm. The participants were asked screening questions regarding age and the city of residence to ensure that only Nashville residents with age over 18 years participate in the survey. The respondents were also asked about the five-digit ZIP code of the home location, which was subsequently used to generate synthetic population and social networks (see Sections 5.1 and 3.3.6 for details). To ask for such detailed residential information, we had to take approval from the Institutional Review Board (IRB) at the University of Memphis under the "*Expedited*" track. The Zipcode-level spatial distribution of the respondents in Nashville is displayed in Fig. 4.

Apart from the DCE and two WOM experiments, the survey asked respondents about their demographic characteristics, accident history, and social ties (see Figure A.1 in the supplementary material for the definition of a close social tie). We also asked respondents to report their perceptions about independent statements on the five-point Likert scale – the indicators of the respondent's anticipated risk in adopting AVs (exact question wording is provided in Fig. 6 in Section 2.4). To maintain the sample quality, participants with a response time below 50% of the median response time were removed - a standard practice in the marketing literature (Callegaro et al., 2014; Greszki et al., 2015; Roßmann, 2010). We removed 106 fast responses, and the final sample had 1,495 complete responses. The marginal distributions of sample and population (as per American Community Survey 2017 obtained from (Manson et al., 2019))

#### S. Dubey et al.

#### Table 1

Attribute levels in the discrete choice experiment and experiment design to capture word of mouth (WOM) effect related to the adoption of autonomous cars.

Attribute	Alternatives Conventional car	Autonomous car
	ALLAD.	
Discrete choice experiment	<b>D</b> (11) (1) (1)	
Purchase price (US\$)	Reported by the respondent	1 20% higher than the cost of conventional car
		2 30% higher than the cost of conventional car 2 40% higher than the cost of conventional car
		4 50% higher than the cost of conventional car
% of people in respondent's social network who adopted autonomous cars	Not applicable	1 20%
% of people in respondent's social network who adopted autonomous cars	Not applicable	2 60%
		3 90%
% of people in respondent's city who adopted autonomous cars	Not applicable	1 30%
is of people in respondent b erry who adopted autonomous cars	Not applicable	2 60%
		3 90%
WOM Experiment 1: safety aspects of autonomous cars		
Source of information	1 Friend	Same as the one for conventional car
	2 Car dealer	
	3 Colleague	
	4 Media	
Vehicle crashes per 100 million miles	1,090	1 415
*		2 290 (30% less than 415)
		3 207 (50% less than 415)
Crashes with no clarity on responsibility/liability	Not applicable	1 10%
		2 30%
WOM Experiment 2: environmental friendliness, travel time savings, and	safety aspects of autonomous	cars
Source of information	1 Friend	Same as the one for conventional car
	2 Car dealer	
	3 Colleague	
	4 Media	
Travel time reduction in autonomous cars	Not applicable	1 20% less than conventional cars
		2 40% less than conventional cars
CO <sub>2</sub> emissions reduction in autonomous cars	Not applicable	1 30% less than conventional cars
		2 50% less than conventional cars
Crashes with no clarity on responsibility/liability	Not applicable	1 10%
		2 30%

across ethnicity, gender, and age are shown in Fig. 5. The overall sampling distribution across all demographic levels is comparable to the population distribution.

## 3.4. Summary Statistics

Table 2 summarizes the descriptive statistics of the survey sample. From a social network perspective, respondents have about 12 close social ties. Among the individuals with accident history, most of them incurred minor damages, and a minority of the sample (6%) suffered from severe injuries.

The distributions of responses to the statements indicating the risk perception of the respondent on the five-point Likert scale are shown in Fig. 6. Over half of respondents are worried about the value-for-money in an AV purchase (Ind01). Only 27% of the respondents are willing to take a risk of purchasing an AV to have an exciting experience (Ind02). 58% of the respondents indicated that they would be uncomfortable in switching to AVs (Ind03), but this might not be specific to automation technology because around 45% of respondents generally struggle with such risky decisions (Ind04).

## 4. Consumer Behavior Model

#### 4.1. Word of Mouth (WOM)

There is well-documented evidence of social influence in the purchase of ice cream (Richards et al., 2014), electronic equipment (Narayan et al., 2011), smartphone (Park and Chen, 2007), organic food items (Chen, 2007), and automobile (Grinblatt et al., 2008). The literature classifies social influence into three major outcomes – conformity, compliance, and obedience (Maness et al., 2015). Conformity is the most common social influence that occurs when an individual changes the behavior to gain acceptance in a group or improve social status by impressing others. Compliance and obedience occur in situations where an individual is requested and ordered to change the behavior, respectively. Conformity is the most plausible form of social influence in the context of AVs and other

Suppose your friend provides you with the following information on the crash and fatality rates of conventional and autonomous cars if both vehicles are driven for <u>100 million miles</u>:

Conventional cars		Autonom	ious cars
Crashes	Fatalities	Crashes	Fatalities
1,090	1.13	290	0.00

The above table indicates that autonomous cars have a much lower crash and fatality rates, but **your friend** also informs that it is not clear who is responsible for the crash in **30**% of crashes encountered by **autonomous cars**.

Assume people in your close social network ask about your opinion on buying an **autonomous car over a conventional** car. Based on the information presented above, to what extent do you agree the following statements:

	Strongly agree	Somewhat agree	Neutral	Somewhat disagree	Strongly disagree
I will suggest them to <b>consider buying an autonomous car</b> over a conventional car because the former is much <b>safer</b> .	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
I will be <b>neutral</b> in my recommendation.	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
I will suggest them to consider buying a conventional car over an autonomous car because at least one knows who is responsible for a crash in a conventional car.	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$

Fig. 2. An example scenario that captures WOM related to the safety of autonomous cars (experiment 1).

novel products. Our review suggests that there could be four ways to model information transfer through WOM and conformity behavior.

The *first* approach is based on the threshold-based specification, which defines a threshold on the proportion of population and friends who must adopt the product before the individual does so (Granovetter, 1978). Such effects need to be incorporated in AV adoption because previous studies on EVs have found the presence of threshold effects. For instance, Mau et al. (2008) observed a significant increase in the individual's willingness to pay for EVs as a function of overall market share. Threshold effects can be incorporated by explicitly modeling a willingness to adopt innovation as a function of market share (Zhang et al., 2011), or an additive utility component can be added based on the choice of others in the network (He et al., 2014; Hsu et al., 2013; Kim et al., 2014; Rasouli and Timmermans, 2016). The *second* approach considers that the exact nature of social influence is unknown, and the total utility of the product is comprised of an individual's utility plus a weighted sum of utilities of others in the social network (see Anselin (2013) and Bhat (2014) for applications in spatial econometrics). The *third* and *fourth* approaches assume that individuals revise their attribute importance weight and attribute value itself as a function of preferences of others in their interpersonal network (Narayan et al., 2011).

While threshold-based specification is straightforward to incorporate, they mask the propagation of information in the social network and only offer an aggregate behavioral effect. In the absence of such micro-level dynamics, the interplay between WOM and risk preferences is difficult to model. From the econometric perspective, this approach could lead to biased parameter estimates as it ignores the spatial/social-network-level correlations among individuals. The last two approaches are information hungry as they require precise information about attribute level communication within the social network. Therefore, we mainly rely on the second approach to model social network effects and capture aggregate threshold effects by controlling for the adoption rates of AVs within the individual's social network and the city.

Generally, the exact nature of social influence is not discernable in the second approach, especially when the social network effect is directly incorporated in the utility equation of a choice model (Bhat et al., 2015; Sidharthan and Bhat, 2012). Moreover, such specification does not offer any means for information propagation without directly including several information-related indicators in the utility equation. These additional indicators could induce measurement error and increase the number of covariates substantially. On the other hand, applying the second approach through the latent construct of the ICLV structure has two advantages. First, the essential information related to social influence (i.e., attitudes such as risk-aversion and positive WOM dissemination) can be encapsulated in a low-dimensional vector of latent variables. This specification is also much less vulnerable to measurement errors. Second, the analyst can incorporate the social network effect on latent variables to enable the exchange of information between consumers, a critical trait of a consumer behavior model that makes it useful for forecasting the adoption of novel products (Bhat et al., 2016b). Thus, the ICLV model helps open the black box by putting a structure on the information dissemination (see Bhat et al. (2015, 2016b) for ICLV applications). Such behavioral insights derived from ICLV make it an attractive alternative to model social influence. In this study, the

Suppose a print or electronic media channel provides you with the following information on the travel time and emission reduction in autonomous cars as compared to conventional cars if both vehicles have an equal market share (i.e., 50% market penetration of autonomous cars): Autonomous cars Reduction in travel time Reduction in CO<sub>2</sub> emissions 20% 30% But the media channel also informs you that there are two potential limitations of autonomous cars: . Whereas you can accelerate the conventional car to reach on time for important meetings or flights, you will not have such flexibility to make the last moment decision while riding autonomous cars. 2 During an accident, about 30% of the time it will not be clear who is responsible for the crash encountered by autonomous cars. Assume people in your close social network ask about your opinion on buying an autonomous car over a conventional car. Based on information presented above, to what extent do you agree to the following statements: Somewhat Somewhat Strongly Strongly agree Neutral disagree disagree agree I will suggest them to consider buying an autonomous car over a conventional car because the former is more reliable and environmental-friendly. I will be neutral in my recommendation. I will suggest them to consider buying a conventional car over an autonomous car because at least one can make ast-moment decisions to accelerate and who is esponsible for a crash in a conventional car.

Fig. 3. An example scenario that captures WOM related to environmental friendliness, travel time savings, and safety aspects of autonomous cars (experiment 2).

information spread by an individual is characterized by the WOM latent variable, which is a function of the WOM of individuals in the social network (see mathematical details in Section 3.3). WOM is measured using responses to indicator questions presented in the WOM experiments (see Fig. 2 and Fig. 3).

## 4.2. Risk Preference

Risk can be broadly classified into seven types – financial, performance, physical, time, social, psychological, and network externality (see Dholakia (2001); Hirunyawipada and Paswan (2006) for a detailed review). By measuring the risk aversion latent variable through the responses to four risk-related statements summarized in Fig. 6, ICLV accounts for psychological risks associated with AV adoption. By directly incorporating AV adoption rates at the social network and city level in the utility equation of ICLV, we implicitly model the perceived time risk. The information provided in WOM experiments regarding the lack of liability in AV crashes accounts for performance risk, and dependence of an individual's WOM on the WOM of social network captures social risk. Thus, the proposed interdependent ICLV model could capture *performance* and *social* risks with the cross-loading of WOM on the risk aversion latent variable.



Fig. 4. Spatial distribution of survey respondents in Nashville.



Fig. 5. Marginal distribution of sample and population across demographics.

#### Table 2

Descriptive statistics of the sample.

Categorical variables				
Variable	Percentage	Variable	Percentage	
Gender		Any kind disability undermining ability to dri	ve	
Male	40%	Yes	11%	
Female	60%	No	89%	
Age		Ethnicity		
18 to 35 years	40%	White	78%	
35 to 65 years	46%	African American	14%	
more than 65 years	15%	Others	8%	
Educational Attainment		Annual household income		
High school or below	20%	less than \$25,000	16%	
Some College or College graduate	61%	\$25,000-\$35,000	11%	
Master's (MS) or Doctoral Degree (Ph.D.)	15%	\$35,000-\$75,000	35%	
Professional Degree (MD, JD, etc.)	4%	\$75,000-\$125,000	22%	
		More than \$125,000	16%	
Willingness to pay to purchase a new car		Number of children in household		
less than \$15,000	31%	Zero	65%	
\$15,000-\$30,000	40%	One or more	35%	
more than \$30,000	29%			
Involve in accidents where vehicle incurred minor damages		Involved in accidents where vehicle incurred	major damages	
Yes	56%	Yes	36%	
No	44%	No	64%	
Involved in accidents and suffered from minor injuries		Involved in accidents and suffered from severe injuries		
Yes	24%	Yes	6%	
No	76%	No	94%	
Continuous variables				
Number of vehicles in the household		Number of workers in the household		
Mean	2.82	Mean	1.58	
Standard deviation	0.93	Standard deviation	0.66	
Household members		Number of social ties		
Mean	2.53	Mean	11.66	
Standard deviation	1.03	Standard deviation	61.22	
Household's frequency of purchasing a car (in years)				
Mean	6.98			
Standard deviation	4.47			



Fig. 6. Descriptive statistics of statements related to risk perception associated with the autonomous car adoption.



Fig. 7. Modeling framework of the Interdependent ICLV model.

## 4.3. Interdependent Integrated Choice and Latent Variable Model (ICLV)

Fig. 7 details the three components of the interdependent ICLV model – structural equation model, measurement equation model, and discrete choice model. The figure highlights how the WOM of decision-makers is affected by the information obtained from external sources and their interpersonal network. The combined information (represented by WOM) is considered to influence the risk-aversion behavior of decision-makers. Finally, information and risk-aversion attitude, along with product attributes, determine the AV adoption behavior.

We provide a general formulation of all three components of the model and discuss them in the context of the empirical study. Subsequently, we write the joint likelihood function and discuss the estimation details. The methodology is based on the ICLV model of Bhat et al. (2016b), but we make two extensions to the existing model. First, we capture the moderation effect between attitudes in the structural equation through cross-loading of latent variables. Second, the discrete choice component of the ICLV model is adjusted to account for panel effects. We write our own code in GAUSS, a matrix programming language, to estimate the interdependent ICLV model.

#### 4.3.1. Latent Variable Structural Equation Model

Let *l* and *q* be the indexes for latent variables l = (1, 2, ..., L) and individuals q = (1, 2, ..., Q). In the empirical study, we consider WOM (l = 1) and risk aversion (l = 2) as two latent variables (i.e., L = 2). The latent variable ( $z_{ql}^*$ ) is written as a linear function of covariates:

$$z_{al}^* = \boldsymbol{\alpha}_l \, \boldsymbol{s}_{ql} + \boldsymbol{\eta}_{ql} \tag{1}$$

Eq. (1) assumes that the individual's latent attitude is independent of other individuals. To incorporate the effect of interpersonal network, Eq. (1) is modified as follows:

$$z_{ql}^{*} = \boldsymbol{\alpha}_{l} \, \boldsymbol{s}_{ql} + \eta_{ql} + \delta_{l} \sum_{q'=1}^{Q} w_{qq'} \, z_{q'l}^{*}$$
<sup>(2)</sup>

where  $s_{ql}$  is a ( $F \times 1$ ) vector of observed covariates,  $\alpha_l$  is the corresponding vector of coefficients,  $\eta_{ql}$  is a normally distributed error term,  $\delta_l$  ( $0 \le \delta_l \le 1$ ) is the autoregressive parameter which captures the interdependence effect across individuals in the interpersonal network, and  $w_{qq'}$  is the weight associated with q and q' with  $w_{qq} = 0$ ;  $w_{qq'} = w_{q'q}$  and  $\sum_{q'\neq q}^{Q} w_{qq'} = 1 \forall q$ . Essentially, Eq. (2) is a spatial auto-correlation regression (Anselin, 2013). In this study,  $s_{ql}$  consists of demographic characteristics, information variables (both type and source of) and accident history variables. The variables presented in the WOM experiments are only included in equation corresponding to WOM (i.e.,  $s_{q1}$ ). We estimate  $\delta_1$  and set  $\delta_2$  to zero because social network effects in risk aversion are transmitted through cross-loading of WOM. We define the following notations to write Eq. (2) in matrix form for all Q individuals:

$$\begin{aligned} \boldsymbol{z}_{q}^{*} &= \left(\boldsymbol{z}_{q1}^{*}, \boldsymbol{z}_{q2}^{*}, \cdots, \boldsymbol{z}_{qL}^{*}\right) & [L \times 1 \text{ vector}], \\ \boldsymbol{z}^{*} &= \left(\left(\boldsymbol{z}_{1}^{*}\right)^{'}, \left(\boldsymbol{z}_{2}^{*}\right)^{'}, \cdots, \left(\boldsymbol{z}_{Q}^{*}\right)^{'}\right)^{'} & [QL \times 1 \text{ vector}], \\ & \widetilde{\boldsymbol{s}}_{q} &= \left(\boldsymbol{s}_{q1}^{'} & \boldsymbol{0} \\ & \ddots \\ \boldsymbol{0} & \boldsymbol{s}_{qL}^{'}\right)^{'} & [L \times LF \text{ matrix}], \\ & \widetilde{\boldsymbol{s}} &= \left(\boldsymbol{\tilde{s}}_{1}^{'}, \boldsymbol{\tilde{s}}_{2}^{'}, \cdots, \boldsymbol{\tilde{s}}_{Q}^{'}\right)^{'} & [QL \times LF \text{ matrix}], \\ & \boldsymbol{\alpha} &= \left(\boldsymbol{\alpha}_{1}^{'}, \boldsymbol{\alpha}_{2}^{'}, \cdots, \boldsymbol{\tilde{s}}_{Q}^{'}\right)^{'} & [LF \times 1 \text{ vector}], \\ & \boldsymbol{\eta}_{q} &= \left(\boldsymbol{\eta}_{q1}, \boldsymbol{\eta}_{q2}, \cdots, \boldsymbol{\eta}_{qL}\right)^{'} & [LX \times 1 \text{ vector}], \\ & \boldsymbol{\eta} &= \left(\boldsymbol{\eta}_{1}, \boldsymbol{\eta}_{2}^{'}, \cdots, \boldsymbol{\eta}_{Q}^{'}\right)^{'} & [QL \times 1 \text{ vector}], \\ & \boldsymbol{\delta} &= \left(\boldsymbol{\delta}_{1}, \boldsymbol{\delta}_{2}, \cdots, \boldsymbol{\delta}_{L}\right)^{'} & [LX \times 1 \text{ vector}], \\ & \boldsymbol{\tilde{\delta}} &= \mathbf{1}_{Q} \otimes \boldsymbol{\delta} & [QL \times 1 \text{ vector}], \end{aligned}$$

where " $\otimes$ " represents the Kronecker product, and  $\mathbf{1}_Q$  is a vector of size ( $Q \times 1$ ) with all its elements equal to 1. To allow for correlation among the latent variables of an individual,  $\eta_q$  is assumed to follow a multivariate normal (MVN) distribution  $\eta_q \sim \text{MVN}_L[\mathbf{0}_L, \Gamma]$ ,  $\Gamma[\eta_q \sim \text{MVN}_L[\mathbf{0}_L, \Gamma]$ , where  $\mathbf{0}_L$  is an ( $L \times 1$ ) column vector of zeros, and  $\Gamma$  is the correlation matrix of size ( $L \times L$ ). Considering L = 2 in this study only  $\Gamma_{12}$  is estimable.

We assume  $\eta_q$  to be independent across individuals (i.e.,  $Cov(\eta_q, \eta_{q'}) = 0$ ,  $\forall q \neq q'$ ). Thus, Eq. (2) can be written in matrix form for all Q individuals as follows:

$$z^* = \mathbf{S}[\tilde{s}\boldsymbol{\alpha} + \boldsymbol{\eta}] \tag{3}$$

where  $\mathbf{S} = [\mathbf{IDEN}_{QL} - \delta \cdot (\mathbf{W} \otimes \mathbf{IDEN}_L)]^{-1}$  is a matrix of size  $(QL \times QL)$ , ".\*" represents the product of each element of a vector with the corresponding row of a matrix,  $\mathbf{IDEN}_{QL}$  is an identity matrix of size  $(QL \times QL)$ ,  $\mathbf{IDEN}_L$  is an identity matrix of size  $(L \times L)$ , and  $\mathbf{W}$  is a symmetric  $(Q \times Q)$  row normalized weight matrix.

While the attitude of individuals is affected by their interpersonal/spatial network, there may also be a moderation effect across different types of attitudes. For example, if an individual has a strong sense of duty towards the environment (pro-environment), he/ she may also exhibit a strong attitude towards trying new environmental-friendly products even if they are novel in the market (risk-taking behavior). Similarly, in the context of this study, WOM can alleviate or augment the risk aversion. Therefore, we extend Eq. (3) to accommodate moderation effect as follows:

We define an indicator matrix **L**-mat of size  $(L \times L)$  with zeros in all cells. If the analyst wishes to load latent variable l' on l', a value of "1" is inserted in the cell (l', l') of the matrix **L**-mat. For two latent variables in the current study where WOM (l = 1) is loaded on the

risk aversion (*l* = 2), the matrix **L**-mat can be written as follows:  $\mathbf{L}$ -mat =  $\begin{bmatrix} 0 & 0 \\ 1 & 0 \end{bmatrix}$ . Next, we define a matrix **R** of size (*L* × *L*) with zeros in all cells and place the corresponding cross-loading parameter [ $\boldsymbol{\rho} = (\rho_1, \rho_2, ..., \rho_{(L^*(L-1)^*0.5) - L})' [\{(L^*(L-1)^*0.5) - L\} \times 1 \text{ vector}]\}$  in the

cells of matrix **R** based on the following pseudocode:

$$\mathbf{R} = zeros(L,L)$$
  

$$count = 0$$
  

$$for i = 1 to L$$
  

$$for j = 1 to L$$
  

$$if(i \neq j)$$
  

$$if(I_{-}[i,j] = 1)$$
  

$$count + = 1$$
  

$$\mathbf{R}[i,j] = \boldsymbol{\rho}[count]$$

The matrix **R** for the case study has the following configuration:  $\mathbf{R} = \begin{bmatrix} 0 & 0 \\ \rho_1 & 0 \end{bmatrix}^2$ . With this information, we extend Eq. (3) to account for moderation effects along with spatial effects:

$$z^* = \mathrm{DS}[s\alpha + \eta] \tag{4}$$

where  $\mathbf{D} = \mathbf{IDEN}_Q \otimes [\mathbf{IDEN}_L - \mathbf{R}]^{-1}$ . Thus,  $z^* \sim \text{MVN}_{QL}(\widetilde{\theta}, \widetilde{\Xi}) z^* \sim \text{MVN}_{QL}(\widetilde{\theta}, \widetilde{\Xi})$ , where mean is  $\widetilde{\theta} = \mathbf{DS}[\widetilde{s}\alpha]$  and correlation matrix is  $\widetilde{\Xi} = \mathbf{DS}[\widetilde{s}\alpha]$ .

<sup>&</sup>lt;sup>2</sup> The loading matrix would be a lower-triangular matrix to ensure one-way cross-loading/structural effect.

S. Dubey et al.

 $\mathbf{D}(\mathbf{S}[\mathbf{IDEN}_Q \otimes \boldsymbol{\Gamma}]\mathbf{S}')\mathbf{D}'.$ 

#### 4.3.2. Latent Variable Measurement Equation Model

Since all indicators to capture the underlying attitudes are measured on Likert scale, we only present the measurement equation system corresponding to ordinal variables. Readers are referred to Bhat et al. (2016a) for a comprehensive measurement equation system with a combination of continuous, ordinal, count, and nominal outcomes.

Let *h* be the index for ordinal variables h = (1, 2, ..., H) and j = (1, 2, ..., J) be the number of categories for ordinal outcomes. In this study, H = 7 (3 for WOM [see Fig. 3] and 4 for risk aversion [see Fig. 6]) and J = 5. Let  $y_{qh}^*$  be the latent variable which leads to the observed outcome  $y_{qh}$  for individual *q* and ordinal variable *h*. Following the usual ordered response formulation, we can write the link function:

$$y_{qh}^* = \gamma_h x_{qh}^* + d_h z_q^* + \varepsilon_{qh}, \quad \psi_{h,y_{qh}-1} < y_{qh}^* < \psi_{h,y_{qh}}$$

$$\tag{5}$$

where  $\mathbf{x}_{qh}^*$  is a ( $K \times 1$ ) vector of observed covariates (including the constant),  $\gamma_h$  is the corresponding vector of coefficients,  $\mathbf{d}_h$  is a ( $L \times 1$ ) vector of latent variable loadings on the ordinal variable h,  $\varepsilon_{qh}$  is a normally distributed random error term, and  $\psi_{h,y_{qh}}$  is the threshold. For each ordinal variable, the thresholds should be in ascending order and should cover the real line, i.e.  $\psi_{h,0} < \psi_{h,1} < \psi_{h,2}$ .  $\dots < \psi_{h,J-1} < \psi_{h,J}$ , where  $\psi_{h,0} = -\infty$  and  $\psi_{h,J} = \infty$ . To set the origin of the ordinal variable, one can either set the second threshold ( $\psi_{h,1}$ ) or the intercept to zero. Here, we choose to do the former:  $\psi_{h,1} = 0$ . In the empirical study,  $\mathbf{x}_{qh}^*$  only has a constant (K = 1). Next, we define the following notations to write Eq. (5) in a matrix form.

$$\begin{aligned} \mathbf{y}_{q}^{*} &= \left(\mathbf{y}_{q1}^{*}, \mathbf{y}_{q2}^{*}, \dots, \mathbf{y}_{qH}^{*}\right)^{'} [H \times 1 \text{ vector}], \\ \mathbf{y}^{*} &= \left(\left(\mathbf{y}_{1}^{*}\right)^{'}, \left(\mathbf{y}_{2}^{*}\right)^{'}, \dots, \left(\mathbf{y}_{Q}^{*}\right)^{'}\right)^{'} [QH \times 1 \text{ vector}], \\ \mathbf{x}_{q}^{*} &= \begin{pmatrix} \left(\mathbf{x}_{q1}^{*}\right)^{'} & 0 \\ \ddots \\ 0 & \left(\mathbf{x}_{qH}^{*}\right)^{'} \end{pmatrix} [H \times HK \text{ matrix}], \\ \tilde{\boldsymbol{\gamma}} &= \left(\boldsymbol{\gamma}_{1}, \boldsymbol{\gamma}_{2}, \dots, \boldsymbol{\gamma}_{H}\right)^{'} [HK \times 1 \text{ vector}], \\ \mathbf{d} &= \begin{pmatrix} \mathbf{d}_{1}^{'} \\ \vdots \\ \mathbf{d}_{H}^{'} \end{pmatrix} [H \times L \text{ matrix}], \\ \tilde{\boldsymbol{\varepsilon}}_{q} &= \left(\varepsilon_{q1}, \varepsilon_{q2}, \dots, \varepsilon_{qH}\right)^{'} [H \times 1 \text{ vector}], \\ \boldsymbol{\psi}_{h} &= \left(\boldsymbol{\psi}_{h,0}, \boldsymbol{\psi}_{h,1}, \dots, \boldsymbol{\psi}_{h,J}\right)^{'} [J \times 1 \text{ vector}], \\ \boldsymbol{\psi}_{q,low} &= \left(\boldsymbol{\psi}_{1,yq_{1}-1}, \boldsymbol{\psi}_{2,yq_{2}-1}, \dots, \boldsymbol{\psi}_{H,yq_{H}-1}\right)^{'} [H \times 1 \text{ vector}], \\ \boldsymbol{\psi}_{q,up} &= \left(\boldsymbol{\psi}_{1,yq_{1}}, \boldsymbol{\psi}_{2,yq_{2}}, \dots, \boldsymbol{\psi}_{Q,up}\right)^{'} [QH \times 1 \text{ vector}], \\ \boldsymbol{\psi}_{low} &= \left(\boldsymbol{\psi}_{1,low}, \boldsymbol{\psi}_{2,low}, \dots, \boldsymbol{\psi}_{Q,up}\right)^{'} [QH \times 1 \text{ vector}], \\ \boldsymbol{\psi}_{up} &= \left(\boldsymbol{\psi}_{1,up}, \boldsymbol{\psi}_{2,up}, \dots, \boldsymbol{\psi}_{Q,up}\right)^{'} [QH \times 1 \text{ vector}], \end{aligned}$$

Eq. (5) can be written in matrix form for individual q as follows:

$$\mathbf{y}_{q}^{*} = \mathbf{x}_{q}^{*} \mathbf{\gamma} + d\mathbf{z}_{q}^{*} + \mathbf{\varepsilon}_{q}, \boldsymbol{\psi}_{q,low} < \mathbf{y}_{q}^{*} < \boldsymbol{\psi}_{q,up} \tag{6}$$

To reduce the model complexity,  $\varepsilon_q$  is assumed to follow a standard MVN distribution:  $\varepsilon_q \sim \text{MVN}_{H \times H}(\mathbf{0}_H, \mathbf{IDEN}_H)\varepsilon_q \sim \text{MVN}_{H \times H}(\mathbf{0}_H, \mathbf{IDEN}_H)$  because the latent variables  $(z_q^*)$  loadings naturally generate the correlation across ordinal variables.

#### 4.3.3. Discrete Choice Model

Let t = (1, 2, ..., T) be the index for choice occasion and i = (1, 2, ..., I) be the index for alternative. In the case study, T = 3 and I = 2. The indirect utility of individual q due to choosing alternative i during choice occasion t is:

$$U_{qti} = \boldsymbol{b}_{i} \boldsymbol{x}_{qti} + \boldsymbol{\lambda}_{i} \boldsymbol{z}_{q}^{*} + \boldsymbol{\vartheta} \left( \boldsymbol{A}_{qti} \boldsymbol{z}_{q}^{*} \right) + \boldsymbol{\zeta}_{qti}$$

$$\tag{7}$$

where  $\mathbf{x}_{qti}$  is an ( $M \times 1$ ) vector of attributes (including the constant),  $\mathbf{b}_i$  is a vector of corresponding marginal utilities,  $\lambda_i$  is a ( $L \times 1$ ) vector of coefficients of latent variables for alternative *i*, and  $\varsigma_{ati}$  is a normally-distributed error term. In the empirical study,  $\mathbf{x}_{ati}$ 

consists of alternative-specific constant, demographic variables, purchase price, percentage adoption in the social network, and percentage adoption in the city. Note that  $\lambda_i$  for one alternative is normalized to zero for identification.  $A_{qti}$  is a ( $LM \times L$ ) matrix of alternative-specific attributes ( $\mathbf{x}_{qti}$ ) that interact with latent variables, and  $\vartheta$  is the corresponding ( $LM \times 1$ ) column vector of marginal utilities. We define the following notations to convert Eq. (7) into matrix form:

$$\begin{split} \boldsymbol{U}_{qt} &= \left(\boldsymbol{U}_{q1}, \boldsymbol{U}_{q2}, ..., \boldsymbol{U}_{qt}\right)^{\boldsymbol{\cdot}} [I \times 1 \text{ vector}], \\ \boldsymbol{U}_{q} &= \left(\boldsymbol{U}_{q1}^{\prime}, \boldsymbol{U}_{q2}^{\prime}, ..., \boldsymbol{U}_{qT}^{\prime}\right)^{\boldsymbol{\cdot}} [TI \times 1 \text{ vector}], \\ \boldsymbol{U} &= \left(\boldsymbol{U}_{1}^{\prime}, \boldsymbol{U}_{2}^{\prime}, ..., \boldsymbol{U}_{Q}^{\prime}\right)^{\boldsymbol{\cdot}} [QTI \times 1 \text{ vector}] \\ \boldsymbol{X}_{qt} &= \left(\begin{matrix} \mathbf{x}_{q1}^{\prime} & 0 \\ \ddots & \mathbf{x}_{qtl} \end{matrix}\right)^{\boldsymbol{\cdot}} [I \times IM \text{ matrix}], \\ \boldsymbol{X}_{q} &= \left(\begin{matrix} \mathbf{x}_{q1}^{\prime} \\ \mathbf{x}_{q2}^{\prime} \\ \vdots \\ \mathbf{x}_{qT} \end{matrix}\right)^{\boldsymbol{\cdot}} [I \times IM \text{ matrix}], \\ \boldsymbol{\tilde{b}} &= \left(\boldsymbol{b}_{1}^{\prime}, \boldsymbol{b}_{2}^{\prime}, ..., \boldsymbol{b}_{I}^{\prime}\right)^{\boldsymbol{\cdot}} [IM \times 1 \text{ vector}], \\ \boldsymbol{\tilde{b}} &= \left(\boldsymbol{b}_{1}^{\prime}, \boldsymbol{b}_{2}^{\prime}, ..., \boldsymbol{b}_{I}^{\prime}\right)^{\boldsymbol{\cdot}} [IM \times 1 \text{ vector}], \\ \boldsymbol{\tilde{b}} &= \left(\boldsymbol{b}_{1}^{\prime}, \boldsymbol{b}_{2}^{\prime}, ..., \boldsymbol{b}_{I}^{\prime}\right)^{\boldsymbol{\cdot}} [I \times L \text{ matrix}], \\ \boldsymbol{\tilde{b}} &= \left(\boldsymbol{b}_{1}^{\prime}, \boldsymbol{b}_{2}^{\prime}, ..., \boldsymbol{b}_{I}^{\prime}\right)^{\boldsymbol{\cdot}} [I \times L \text{ matrix}], \\ \boldsymbol{\tilde{b}} &= \left(\boldsymbol{b}_{1}^{\prime}, \boldsymbol{b}_{2}^{\prime}, ..., \boldsymbol{b}_{I}^{\prime}\right)^{\boldsymbol{\cdot}} [I \times L \text{ matrix}], \\ \boldsymbol{\tilde{b}} &= \left(\boldsymbol{b}_{1}^{\prime}, \boldsymbol{b}_{2}^{\prime}, ..., \boldsymbol{b}_{I}^{\prime}\right)^{\boldsymbol{\cdot}} [I \times L \text{ matrix}], \\ \boldsymbol{\tilde{b}} &= \left(\boldsymbol{b}_{1}^{\prime}, \boldsymbol{b}_{2}^{\prime}, ..., \boldsymbol{b}_{I}^{\prime}\right)^{\boldsymbol{\cdot}} [I \times L \text{ matrix}], \\ \boldsymbol{\tilde{b}} &= \left(\boldsymbol{b}_{1}^{\prime}, \boldsymbol{b}_{2}^{\prime}, ..., \boldsymbol{b}_{I}^{\prime}\right)^{\boldsymbol{\cdot}} [I \times L \text{ matrix}], \\ \boldsymbol{\tilde{b}} &= \left(\boldsymbol{b}_{1}^{\prime}, \boldsymbol{b}_{2}^{\prime}, ..., \boldsymbol{c}_{qt}\right)^{\boldsymbol{\cdot}} [I \times L \text{ matrix}], \\ \boldsymbol{\tilde{b}} &= \left(\boldsymbol{b}_{1}^{\prime}, \boldsymbol{b}_{2}^{\prime}, ..., \boldsymbol{c}_{qtl}\right)^{\boldsymbol{\cdot}} [I \times L \text{ matrix}], \\ \boldsymbol{\boldsymbol{\sigma}} &= \left(\boldsymbol{\boldsymbol{\sigma}}_{q1}^{\prime}, \boldsymbol{\boldsymbol{\sigma}}_{q2}^{\prime}, ..., \boldsymbol{c}_{qtl}\right)^{\boldsymbol{\cdot}} [I \times I \text{ vector}], \\ \boldsymbol{\boldsymbol{\sigma}} &= \left(\boldsymbol{\boldsymbol{\varsigma}}_{q1}, \boldsymbol{\boldsymbol{\varsigma}}_{q2}, ..., \boldsymbol{\varsigma}_{qtl}\right)^{\boldsymbol{\cdot}} [I \times I \text{ vector}], \\ \boldsymbol{\boldsymbol{\varsigma}} &= \left(\boldsymbol{\boldsymbol{\varsigma}}_{q1}, \boldsymbol{\varsigma}_{q2}, ..., \boldsymbol{\varsigma}_{qtl}\right)^{\boldsymbol{\cdot}} [TI \times 1 \text{ vector}], \end{aligned}$$

Let  $\Lambda$  be the covariance matrix of  $\varsigma_{qt}$ , which we assume to be independent across choice occasions to reduce the model complexity because latent variable ( $z_q^*$ ) loadings generate the correlations across choice occasions. Thus, Eq. (7) for individual q can be written as follow:

$$\boldsymbol{U}_{q} = \boldsymbol{x}_{q} \widetilde{\boldsymbol{b}} + \left(\widetilde{\boldsymbol{\lambda}} + \boldsymbol{\varpi}_{q}\right) \boldsymbol{z}_{q}^{*} + \boldsymbol{\varsigma}_{q}, \tag{8}$$

where  $\varsigma_q \sim MVN_{TI \times TI}(\mathbf{0}, \mathbf{IDEN}_T \otimes \Lambda)$ .

## 4.3.4. Joint Likelihood

Note that obtaining the marginal distribution of  $U_q$  and  $y_q^*$  is not straightforward due to spatial correlation in the latent variable  $(z_q^*)$  across individuals. Therefore, we work with the joint distribution of all individuals to write the model likelihood:

 $\mathbf{y}^* \mathbf{U} \sim MVN_{[\mathcal{Q}(H+TI)] \times [\mathcal{Q}(H+TI)]} (\mathbf{B} = \boldsymbol{\mu}_1 + \boldsymbol{\mu}_2 \widetilde{\boldsymbol{\theta}}, \boldsymbol{\Omega} = \boldsymbol{\mu}_2 \widetilde{\Xi} \boldsymbol{\mu}_2' + \mathbf{IDEN}_Q \otimes \boldsymbol{\Sigma}),$ 

where

$$\boldsymbol{\mu}_{1q} = \left( \begin{bmatrix} \boldsymbol{x}_{q}^{*} \tilde{\boldsymbol{\gamma}} \end{bmatrix}^{'}, \begin{bmatrix} \boldsymbol{x}_{q} \tilde{\boldsymbol{b}} \end{bmatrix}^{'} \right)^{'} [(H + TI) \times 1 \text{ vector}],$$
  
$$\boldsymbol{\mu}_{1} = \left( \boldsymbol{\mu}_{11}^{'}, \cdots, \boldsymbol{\mu}_{1Q}^{'} \right)^{'} [Q(H + TI) \times 1 \text{ vector}],$$
  
$$\boldsymbol{\mu}_{2q} = \left( \boldsymbol{d}^{'}, \left( \tilde{\boldsymbol{\lambda}} + \boldsymbol{\varpi}_{q} \right)^{'} \right)^{'} [(H + TI) \times L \text{ matrix}],$$
  
$$\boldsymbol{\mu}_{2} = \begin{pmatrix} \boldsymbol{\mu}_{21} & 0 \\ 0 & \boldsymbol{\mu}_{2Q} \end{pmatrix} [Q(H + TI) \times QL \text{ matrix}],$$
  
$$= \begin{pmatrix} \mathbf{IDEN}_{H} & \mathbf{0}_{H \times TI} \\ \mathbf{0}_{TI \times H} & \mathbf{IDEN}_{T} \otimes \boldsymbol{\Lambda} \end{pmatrix} [(H + TI) \times (H + TI) \text{ matrix}].$$

Since, only the difference in utility matters, we work with utility differences in the discrete choice part. We specifically subtract the utility of the chosen alternative from utilities of all non-chosen alternatives. Moreover, top left element of the differenced error covariance matrix  $(\tilde{\Lambda})$  is fixed to 1 to set the utility scale for identifiability (Keane, 1992). Thus, for *I* alternatives case, only { $I^*(I-1)$  \*0.5} – 1 covariance elements are identifiable. Further, since all the differenced error covariance matrices must originate from the same undifferenced error covariance matrix ( $\Lambda$ ), we specify matrix  $\Lambda$  as follows (Sidharthan and Bhat, 2012):  $\Lambda = \begin{bmatrix} 0 & 0 \\ 0 & \tilde{\Lambda} \end{bmatrix}$ . We also had to transform Eq. (9) in the utility-difference space using a matrix **M**\_**Diff** of size [ $Q(H + T(I-1)) \times Q(H + TI)$ ] as shown in Eq. (10). The procedure to compute **M**\_**Diff** is provided in Algorithm A.1 in Appendix 1. The resulting joint distribution in the utility differenced

space is:

$$\overline{\mathbf{y}}^* \overline{U} \sim MVN_{[\mathcal{Q}(H+T(I-1))] \times [\mathcal{Q}(H+T(I-1))]}} \left(\overline{\mathbf{B}} = \mathbf{M}_{-} \mathbf{Diff} * \mathbf{B}, \overline{\Omega} = \mathbf{M}_{-} \mathbf{Diff} * \boldsymbol{\Omega} * \mathbf{M}_{-} \mathbf{Diff}'\right)$$
(10)

#### 4.3.5. Estimation

Σ

To evaluate the joint likelihood presented in Eq. (10), we need to compute a Q(H + T(I - 1)) dimensional integral involved in the multivariate normal cumulative distribution function (MVNCDF). Despite several advancements in quasi-Monte Carlo (Bhat, 2003), quadrature (Bansal et al., 2021b), analytical methods (Bhat, 2018) to evaluate high-dimensional integrals, computing such a high-dimensional integral at the desired accuracy remains infeasible. Therefore, we make use of the composite marginal likelihood (CML) approach, which define a surrogate likelihood function by taking the product of low dimensional MVNCDF. The surrogate function for Eq. (10) is as follows:

$$L_{CML}(\Theta) = \left\{ \begin{bmatrix} \prod_{q=1}^{Q} \prod_{h=1}^{H} \prod_{q^{'}=q}^{Q} \prod_{h^{'}=1}^{H} \Pr[y_{qh}, y_{q^{'}h^{'}}] \end{bmatrix} \forall \begin{cases} q \neq q^{'} \\ or \\ q = q^{'} and h^{'} > h \end{cases} \right\}$$
$$\times \left\{ \begin{bmatrix} \prod_{q=1}^{Q} \prod_{q^{'}=1}^{Q} \prod_{h=1}^{H} \prod_{t=1}^{T} \Pr[y_{qh}, q^{'}_{tim}] \end{bmatrix} \right\}$$
$$\times \left\{ \begin{bmatrix} \prod_{q=1}^{Q} \prod_{q^{'}=q}^{T} \prod_{i=1}^{Q} \prod_{q^{'}=i}^{T} \Pr[q_{ii_{m}}, q^{'}_{ii_{m}}] \end{bmatrix} \right\} \langle \begin{cases} q \neq q^{'} \\ or \\ q = q^{'} and t^{'} > t \end{cases} \right\}$$
(11)

where  $\Theta = [Vech(\alpha), Vech(\Gamma), \delta, \rho, Vech(\tilde{\gamma}), Vech(d), \psi, \tilde{b}, Vech(\lambda), \vartheta, Vech(\Lambda)]$ , "Vech(.)" vectorizes all the elements of the matrix,  $y_{qh}$  indicates the observed scale for the ordinal variable h by the individual q, and  $q_{ti_m}$  indicates the chosen alternative  $i_m$  at the choice occasion t by the individual q. In Eq. (11), for each pair of individuals q and q', the first, second, and third terms correspond to the pairing of ordinal variables, pairing of ordinal variables with nominal variables, and pairing of nominal variables, respectively. Please note that the highest dimensionality of integration in Eq. (11) is  $(I - 1)^*2$  as compared to Q(H + T(I - 1)) in Eq. (10). Further details about computing surrogate function are provided in Appendix.

The analyst needs to calculate the robust asymptotic covariance matrix, a function of Jacobian and Hessian matrices, to obtain standard errors of the model parameters (Godambe, 1960). The calculation of Jacobian and Hessian matrices for choice models with interdependencies is different from the traditional choice models. Readers are referred to Sidharthan and Bhat (2012) for a discussion on the calculation of the sandwich covariance matrix.

#### 4.3.6. Spatial Weight Matrix Computation

All the data required to estimate the interdependent ICLV model can be obtained from a stated preference survey, except the weight

matrix used in Eq. (2). Essentially, the weight matrix determines the un-moderated weights assigned by individuals to others in their interpersonal network. The weight matrix can be constructed in two ways. The first method relies on the concept that "everything is related to everything else, but near things are more related than distant things" (Tobler, 1970). This concept has been extensively used in spatial econometrics to construct the weight matrix as a function of spatial distance, such as the inverse of distance (Anselin, 2010). On the other hand, the second method uses the concept of homophilic – similarity in terms of demographic attributes, proximity, attitudes, and other behavioral characteristics. This concept is often used in social sciences to model information propagation and online interactions (Bhattacharya and Sarkar, 2021; David-Barrett, 2020).

In the current study, we construct weight matrices using both methods – i) geographical distance (inverse of distance) and ii) homophilic based weight matrices. For the geographical distance-based weight matrix, the strength of a tie between two individuals is negatively proportional to the Euclidian distance between their residential locations. We use the R package "geosphere" to calculate the shortest pair-wise distance between five-digit zip codes of home locations (Hijmans et al., 2017). We compute the homophilic-based index using the Gower dissimilarity index because it can handle both continuous and categorical data on demographic characteristics (Gower, 1971). The Gower dissimilarity index  $w_{qq'}$  is given by:

$$w_{qq'} = \frac{\sum_{k=1}^{K} \omega_k s_{qq'k}}{\sum_{k=1}^{K} \omega_k}$$
(12)

where  $s_{qq'k}$  is the contribution of the variable k in the similarity between individuals q and  $q'(q \neq q'; q, q' \in Q)$  and  $\omega_k$  is the corresponding weight. For the continuous and categorical variables, contributions are calculated using normalized Manhattan and Dice distances, respectively. A value of  $w_{qq'}$  close to zero indicates a strong tie. We consider demographic characteristics like age, gender, ethnicity, educational attainment, and household income to generate the Gower-distance-based weight matrix. The contribution weight  $\omega_k$  for each variable is assumed to be the same.

As discussed earlier, the weight matrix W is a row normalized matrix (right stochastic matrix). Hence, the autoregressive parameter  $\delta_l$  is bounded between 0 and 1. Further, the weights should be the higher between individuals of similar demographic characteristics. Therefore, the weight matrix constructed using Gower index is modified as 1 - W before performing row normalization.

Further, a closer look at Eq. (11) indicates that the likelihood computation requires the joint probability calculation for all pairs of individuals. Assuming that each individual is connected with every other individual is unrealistic and computationally prohibitive. To make the likelihood estimation tractable, one can either follow Tobler's law (i.e., weight beyond a certain distance is zero) or put constraints on the number of effective social ties in an individual's interpersonal network. In this study, we use the latter and consider two configurations with five and ten social ties, i.e., we determine the nearest/closet five and ten ties/individuals based on Euclidian distance and Gower-similarity index  $(1-w_{qq'})$ . Thus, we estimate the interdependent ICLV model with four specifications of the weight matrix and ties – spatial distance with five ties, spatial distance with ten ties, Gower distance with five ties, and Gower distance with ten ties. Eq. (11) can be modified to restrict likelihood computation based on individual's social network as follows:

$$L_{CML}(\Theta) = \left\{ \begin{bmatrix} \prod_{q=1}^{Q} \prod_{h=1}^{H} \prod_{q'\in\{|q_{k}|\}}^{q_{k}\{|q_{k}|\}} \prod_{h'=1}^{H} \Pr[y_{qh}, y_{q'h'}] \end{bmatrix} \forall \begin{cases} q \neq q' \\ or \\ q = q' \text{ and } h' > h \end{cases} \right\}$$

$$\sum_{\substack{L_{CML}(1) \\ L_{CML}(1)}}^{\dots} \times \begin{bmatrix} \prod_{q=1}^{Q} \prod_{q'=q_{k}\{1\}}^{q_{k}\{|q_{k}|\}} \prod_{h=1}^{H} \prod_{t=1}^{T} \Pr[y_{qh}, q'_{ti_{m}}] \end{bmatrix}$$

$$\times \left\{ \begin{bmatrix} \prod_{q=1}^{Q} \prod_{q'=q_{k}\{1\}}^{q_{k}\{|q_{k}|\}} \prod_{i'=t+1}^{T} \Pr[q_{ii_{m}}, q'_{i'i_{m}}] \end{bmatrix} \forall \begin{cases} q \neq q' \\ or \\ i_{m} \end{bmatrix} \right\}$$

$$\times \left\{ \begin{bmatrix} \prod_{q=1}^{Q} \prod_{t=1}^{T} \prod_{q'=q_{k}\{1\}}^{q_{k}\{|q_{k}|\}} \prod_{i'=t+1}^{T} \Pr[q_{ii_{m}}, q'_{i'i_{m}}] \end{bmatrix} \forall \begin{cases} q \neq q' \\ or \\ q = q' \text{ and } t' > t \end{cases} \right\}$$

$$L_{CML}^{\dots}(3)$$

$$L_{CML}^{\dots}(3)$$

$$L_{CML}^{\dots}(3)$$

$$L_{CML}^{\dots}(3)$$

$$L_{CML}^{\dots}(3)$$

$$L_{CML}^{\dots}(3)$$

$$L_{CML}^{\dots}(3)$$

$$L_{CML}^{\dots}(3)$$

where  $|q_s|$  indicates the cardinality of set  $q_s$ ,  $q_s$  contains the neighbors/closet ties of individual q including q, and  $q_s\{1\}$  indicates the first element of the social network. In the current empirical application, a  $|q_s| = 6(self + 5ties)$  corresponding to five ties entails computation of 455 (5*ties*\*91), 140 (5*ties*\*28), and 75 (5*ties*\*15) two-dimensional MVNCDF per individual for expression  $L_{CML}(1)$ ,  $L_{CML}(2)$ , and  $L_{CML}(3)$ , respectively.

# Table 3 Summary of model fit statistics.

Specification Distance measure	Number of ties	CompositeLog- likelihood	Composite likelihoodinformationcriterion (CLIC)	Choice model Composite log- likelihood	Averageprobability of correct prediction (standard deviation in brackets)
Gower	5	-2,730,103.1	-2,734,658.1	-1,32,048.8	0.581 (0.097)
Spatial	5	-2,738,108.8	-2,741,157.9	-1,32,557.8	0.544 (0.058)
Gower	10	-5,440,721.0	-5,458,700.8	-2,64,773.9	0.581 (0.098)
Spatial	10	-5,457,718.3	-5,464,455.8	-2,64,932.6	0.548 (0.060)

## 5. Results of Consumer Behavior Model

## 5.1. Statistical Model fit assessment

As we estimate the ICLV model using the CML approach, we adopt composite likelihood information criterion (CLIC) to compare non-nested models, i.e. models with the same number of ties but different weight matrix configurations (Varin and Vidoni, 2005). CLIC is computed using the following expression:  $[logL_{CML}(\Theta) - tr(J(\Theta)H(\Theta)^{-1})]$ , where  $[logL_{CML}(\Theta)]$  is the composite log-likelihood and tr  $[J(\Theta)H(\Theta)^{-1}]$  is the penalty term (trace of the product of Jacobian and Hessian inverse). The model with higher CLIC is preferred. Table 3 provides the CLIC statistics and other relevant statistics for all the four weight matrix specifications.

The CLIC statistics suggest that the Gower-similarity-based weight matrix configuration better explains the AV preferences. However, a close look at the average probability of correct prediction suggests that the model fit is only marginally sensitive to the configuration of the spatial weight matrix.

#### 5.1.1. Structural Equation Model Results

Table 4 provides the parameter estimates of the structural equation for latent variables. Since the direction of effects does not change much across weight matrix specifications, we present and discuss results for the weight matrix based on Gower distance with five ties. The results of the other three weight matrix configurations (Table A.2 to A.7) and the non-spatial model (Table A.8 to A.10) are provided in the online supplementary material.

Several demographic characteristics, type and source of information, and accident history have a statistically significant effect on WOM dissemination. For instance, bachelor's degree holders are more likely to spread positive information about AVs than those with lower education levels, ceteris paribus. The positive effect of education could be attributed to the fact that highly educated individuals can process more information and thereby be more certain about the long-term benefits of AVs, such as lower accident rates and emissions (Golbabaei et al., 2020; Haboucha et al., 2017; Han et al., 2011; Jansson et al., 2011; Jerit et al., 2006; Knight et al., 2010; Liljamo et al., 2018). Individuals living in high-income households tend to be more active in disseminating positive information than

#### Table 4

Structural equation model parameter estimates for interdependent ICLV model with weight matrix based on Gower distance with five ties.

Explanatory Variables		Coefficient (t-stat)	
		Word of Mouth (WOM)	Risk Aversion
Education Status	Some college degree or below	-0.101 (-1.90)	_
	Professional degree (MD, JD, etc.)	-0.198 (-2.50)	_
	College Graduate	_	0.239 (3.25)
	MS, PhD or Doctoral degree	_	0.239 (3.25)
Household Income Base: >75K	<=35K	-0.280 (-1.66)	_
	36K - 75K	-0.150 (-3.36)	_
Household Configuration	Number of Workers	0.266 (3.59)	0.055 (1.21)
	Number of Children	0.304 (3.58)	0.074 (1.80)
	Respondent Male (Base: Female)	0.010 (1.18)	-0.256 (-2.29)
Vehicle Ownership	Number of Vehicles	-0.132 (-3.37)	_
Information Variables	Number of crashes in AV	-0.046 (-3.36)	_
	Reduction in travel time in AV	_	_
	Reduction in CO2 emission in AV	_	_
	Unclear liability in x% of AV-involved crashes.	-0.637 (-3.07)	_
Information Source Base: Car Dealer	Friend	0.010 (1.04)	_
	Colleague at work	0.028 (1.15)	_
	Media	0.092 (2.81)	_
Past One year accident involvement.	Vehicle incurred minor damages.	-0.062 (-1.98)	_
Number of accidents where	Vehicle incurred major damages.	0.043 (1.65)	_
	I suffered from minor injuries.	-0.143 (-3.05)	_
	I suffered from severe injuries.	0.087 (2.62)	_
Cross Loading ( $\rho$ )	WOM		-1.049 (-5.12)
Spatial Parameter ( $\delta$ )		0.247 (18.77)	0 (fixed)
Correlation between WOM and Risk Aversio	on $(\Gamma_{12})$	0.369 (1.53)	
'' indicates that the parameter was not sig	nificant at a significance level of 0.2 and hence removed		

their low-income counterparts (Bansal et al., 2016). The effect of income on spreading positive WOM could be related to hedonic experience (Paridon et al., 2006). In terms of household configurations, three demographic variables have a positive effect on WOM dissemination. The number of workers in the household has a positive impact on WOM communication. One possible reason for such positive WOM communication could be the self-relocation and parking capability of the AVs, a trait beneficial for large working households (Baron et al., 2021). The number of children in the household is also linked to the positive dissemination of WOM. The results also suggest that male decision-makers are more likely to spread positive WOM compared to their female counterparts (Kim et al., 2019; Liljamo et al., 2018; Zoellick et al., 2019). A higher propensity of men towards AVs could be attributed to status symbols. Owning an AV can convey a sense of symbol or power manifested through a willingness to pay a premium price for new innovative car technology (Wadud and Chintakayala, 2021).

Vehicle ownership (i.e., the number of cars) is negatively related to WOM dissemination (Liljamo et al., 2018). Since vehicle ownership is generally a proxy for driving propensity (Kaneko and Kagawa, 2021), households with higher vehicle ownership may not want to relinquish the pleasure derived from manual driving. Two of the four information variables are found to be statistically significant in explaining the WOM, namely the number of AV-involved crashes and the proportion of such crashes with the lack of liability. Both covariates have an intuitive sign. We include informational variables in the structural equation part of the model so that their propagation effect through the social network can be explicitly captured. This structural relationship also leads to an elegant top-down propagation of the effect of information from latent variables to observed indicators and choices.

Among information sources, information received from friends, colleagues, and media has a positive influence on WOM compared to that of the car dealer. This result might be a consequence of relatively lesser trust in the information provided by dealers. Finally, the severity of injury/damage in accidents has a stimulating effect on WOM dissemination. In the event of minor injury/damage to the individual/vehicle, the effect is negative. However, the effect turns out to be positive in the event of major injury/damage. The results indicate that people with the worst experience of manual driving are more likely to realize the benefits of automated driving from the safety perspective (Menon et al., 2019).

We now discuss the parameter estimates of the structural equation for the second latent variable – risk aversion. Bachelor's degree holders exhibit higher risk aversion compared to their counterparts with lower education (Jung, 2015), perhaps because they are more aware of the risks associated with the adoption of nascent AV technology. Moreover, females with a higher number of workers and children in the household are likely to be highly risk-averse compared to their counterparts. The cross-loading parameter ( $\rho$ ) (i.e., loading of WOM on risk aversion) is negative and statistically significant, suggesting that the increase in positive WOM reduces risk aversion. The result is aligned with intuition. Note that the actual association of demographic characteristics and risk-averse behavior should be derived after considering the indirect effect of these variables on risk aversion through WOM.

The social network effect is captured through an autoregressive parameter ( $\delta$ )in the structural equation of WOM. The autoregressive parameter is significant and has a value of 0.25. This parameter estimate can be interpreted as the weight given by individuals to the information received from their social network. As discussed earlier, we do not estimate the autoregressive parameter for risk aversion because cross-loading of WOM ( $\rho$ )implicitly captures social network effect. Finally, the error correlation ( $\Gamma_{12}$ ) between WOM and risk aversion is 0.37. Its interpretation is much more nuanced than it appears. A closer look at Eq. 4 reveals that the correlation

matrix of latent variables is  $\Xi = \mathbf{D}(\mathbf{S}[\mathbf{IDEN}_Q \otimes \boldsymbol{\Gamma}]\mathbf{S}')\mathbf{D}'$ . Since all the elements of matrix **S** are positive by being auto-regressive parameter boundness, the inner expression  $\mathbf{S}[\mathbf{IDEN}_Q \otimes \boldsymbol{\Gamma}]\mathbf{S}'$  remains positive as the error-correlation matrix  $\boldsymbol{\Gamma}$ has positive elements, i.e., ( $\Gamma_{12}$ ). The off-diagonal elements of matrix **D** are non-positive numbers since the cross-loading of WOM on risk aversion ( $\rho$ ) is negative. Thus, all the cells representing correlation between WOM and risk aversion for both inter and intra individual have negative

entries. Therefore, the implied correlation  $(\Xi)$  between WOM and risk aversion is negative, suggesting that the aggregate effect of unobserved variables on the latent variables is in the opposite direction. Further, a negative implied correlation suggests that WOM and risk aversion are not-orthogonal (not-independent) latent variables. Non-orthogonality implies that a certain amount of common

## Table 5

Measurement equation model parameter estimates for interdependent ICLV model with weight matrix based on Gower distance with five ties,  $\psi_1 = -\infty$ ,  $\psi_2 = 0$ ,  $\psi_6 = \infty$ .

Statement (Five-point Likert scale with labels strongly disagree to strongly	Coefficient (t-st	at)	Coeffi	cient (standaro	d error)	
agree)	Word of Mouth (WOM)	Risk Aversion	Intercept	$\psi_3$	$\psi_4$	Ψ5
I will suggest them to consider buying an autonomous car over a	1.16		1.562	0.986	2.400	3.856
conventional car because the former is much safer.	(3.46)		(6.42)	(0.142)	(0.324)	(0.514)
I will suggest them to consider buying a conventional car over an	-0.335		1.950	0.974	2.007	2.886
autonomous car because at least one knows who is responsible for a crash in a conventional car.	(-3.56)		(10.63)	(0.051)	(0.069)	(0.087)
I am worried that I might not get value-for-money in an autonomous car		0.277	1.519	0.667	1.429	2.431
purchase.		(2.61)	(16.69)	(0.038)	(0.055)	(0.081)
I would take the risk with autonomous car purchase in exchange for an		-0.560	0.800	0.761	1.686	2.566
exciting and novel experience.		(-3.42)	(6.71)	(0.052)	(0.100)	(0.145)
I would feel uncomfortable in switching to autonomous cars.		1.052	2.630	1.155	2.078	3.491
		(21.27)	(10.14)	(0.154)	(0.270)	(0.452)
I struggle in taking risks with such unconventional decisions.		0.616	1.790	0.942	1.866	2.899
		(7.40)	(7.49)	(0.081)	(0.149)	(0.227)

#### Table 6

Choice model parameter estimates for interdependent ICLV model with weight matrix based on Gower distance with five ties (base: will not buy an AV).

Explanatory Variables		Coefficient (t-stat)
Price and adoption variables	Constant	1.000 (31.03)
	Percentage adoption in social ties	1.081 (4.68)
	Percentage adoption in city/community	0.523 (1.98)
	Ratio of AV to CV price	-2.547 (-3.56)
Household Configuration	Number of workers	0.041 (1.96)
	Number of children	-0.088 (-7.08)
	Respondent Male (Base: Female)	0.129 (11.30)
Latent	WOM	0.605 (12.15)
Variable Loading	Risk Averse	-0.612 (-9.81)
Latent	WOM * Percentage adoption in social ties	_
Variable Interaction	Risk Aversion * Percentage adoption in social ties	1.121 (5.43)
	WOM * Percentage adoption in city/community	0.408 (2.37)
	Risk Aversion * Percentage adoption in city/community	_
	WOM * Ratio of AV to CV price	0.663 (9.26)
	Risk Aversion * Ratio of AV to CV price	_
'' indicates that the parameter is not significant at a	significance level of 0.2.	

behavioural traits (attitudes) is measured by both latent variables.

#### 5.1.2. Measurement Equation Model Results

The parameter estimates for the measurement equation of latent variables are provided in Table 5. Two out of three measurement indicators of WOM and all four measurement indicators of risk aversion show association. All the loading parameters have the expected signs. For example, WOM has positive loading on the statement "I will suggest them to consider buying an AV over a CV because the former is much safer" measured on a five-point Likert scale going from strongly disagree to strongly agree. Other loading parameters can be interpreted similarly. Table 5 also provides the estimates for the intercept and thresholds of ordered indicators. Very small standard errors of thresholds indicate that cut-off values are statistically separated from each other at a significance level of 0.05.

### 5.1.3. Choice Model Results

Table 6 summarizes the parameter estimates of the utility equation of the binary choice variable – whether to buy an AV or not. Aligned with the intuition, the effects of social and city-level adoption rates on the individual's likelihood to adopt an AV are positive and statistically significant (Bansal et al., 2016; Bansal and Kockelman, 2018; Sharma and Mishra, 2020). The effect of price also exhibits the expected trend, as the likelihood of buying an AV decrease with an increase in the AV price relative to the CV price. We consider a non-linear specification with an estimable power parameter on the ratio of AV price and CV price. However, the price effect turns out to be linear because the power parameter is not statistically different from 1 (at a significance level of 0.2).

A few demographic variables directly affect the likelihood of choosing an AV. The likelihood of purchasing an AV increases with the number of workers, possibly due to self-relocation and increased productivity during work-related trips (Lavieri et al., 2017; Saeed et al., 2020; Wadud et al., 2016). An increase in the number of children reduces the likelihood of owning an AV. The indirect effect of the number of children on latent variables explains this direct effect. It is interesting to note that while those with the higher number of children are likely to disseminate positive WOM (Sinha et al., 2020), they also have higher risk aversion. Perhaps, the latter effect dominates in direct effect due to severe concerns related to children's safety (Haboucha et al., 2017). Finally, males are estimated to be more likely to purchase an AV than their female counterparts which is in line with the existing literature (Bansal et al., 2016; Hilgarter and Granig, 2020; Hohenberger et al., 2016; Kyriakidis et al., 2015; Payre et al., 2014; Pew Research Center, 2022; Schoettle and Sivak, 2014). Such finding can be attributed to higher propensity of men being the early adopters of AVs (Hilgarter and Granig, 2020; Payre et al., 2014; Schoettle and Sivak, 2014). Furthermore, as compared to men, women are more likely to be concerned about barriers associated with AVs (Kyriakidis et al., 2015; Pew Research Center, 2022) and willing to pay less for buying an AV (Bansal et al., 2016).

The effect of latent variables on the likelihood of buying an AV has an intuitive sign. For instance, individuals with positive WOM dissemination and lower risk aversion have a higher inclination toward buying AVs. Further, we also explore the interaction effects of latent variables and the adoption rates at both social network and city levels. The interaction parameter estimates indicate that the effect of positive WOM dissemination increases with the increase in city-level AV adoption, whereas the negative effect of risk aversion is pacified by the AV adoption in the social network. Finally, the WOM latent variable also has a statistically significant interaction effect with the ratio of AV price to CV price. The positive interaction effect parameter indicates that the negative effect of AV price decreases with the increase in positive WOM dissemination.

While the parameter estimates of the consumer behavior model provide the directional effect of several variables on the likelihood of AV adoption, they cannot be translated into analytical expressions of the forecasted market share of AVs. To this end, we use these parameter estimates and perform an agent-based simulation to understand the market evolution of AV under various scenarios (e.g., change in AV price and reduction in AV-involved crash rates).

#### 6. Agent-Based Model: simulation

#### 6.1. Synthetic Population Generation

To run an agent-based simulation for the entire household population of Nashville (Davidson County), we expand the survey sample to the household-level synthetic population using an iterative proportional updating (IPU) algorithm. The algorithm adjusts and reallocates weights among households until household- and person-level attributes are both matched with the marginal distributions of attributes in the population (Konduri et al., 2016). We obtain population-level marginal distribution (at household- and person-level) for Nashville from American Community Survey 2013-2017 (Manson et al., 2019). We consider household size, income, and the number of workers, children, and vehicles as the household-level control variables. In addition, age, gender, ethnicity, educational attainment, and disability are used as person-level control variables. To match the spatial distribution of the population, we utilize the five-digit ZIP code of the survey respondent's home location and census tract from the population using their crosswalk (Din and Wilson, 2020). We apply the IPU algorithm through open-source Python software, PopGen (Konduri et al., 2016). The survey sample of 1,495 respondents is thus expanded to a synthetic population of 421,223 households.

### 6.2. Agent-Based Simulation Framework

Fig. 8 presents the agent-based simulation (ABM) framework, which takes the estimated consumer behavior model, synthetic population, agent-level social network, and the control variables mentioned in Table 7 as inputs. The ABM is run for 50 iterations where the iteration corresponds to one year. The AV price at time *t* is obtained using the following discounting equation: starting price  $\times$  (1 – yearly discount rate)<sup>(t-1)</sup>. The two variables related to AV safety and regulation – the number of AV-involved crashes and the proportion of AV-involved crashes with unknown liability – are generated based on their range in the WOM choice experiment. The magnitude of both variables is iteratively reduced to represent the expectations of experts that AV-involved crashes would reduce with the advancement in the automation technology and stronger laws will be developed to ensure liability of AV-involved crashes. Specifically, we adopt the following quadratic function to find the value of these variables at time *t* (see Table 1 for upper limits):Upperlimit – ( $\sqrt{Upperlimit}$ ) \* (t/Curvature) – 1. Moreover, the percentage of the population receiving the information about the above-discussed two information variables through media in year *t* is determined by the following function:  $2\sqrt{t} + 2t + 1$ . This function ensures an increase in media exposure over time. For example, the function implies that 5% of people receive information through media in the first year, but the proportion jumps to 15% in the fifth year and 72% in the thirtieth year. Finally, an agent also gets information from a *friend* when the AV adoption in his/her social network exceeds 40%.

By inserting the above-discussed inputs, spatial weight matrix, agent's social network, and other characteristics of agents in Eq. (4), we calculate the values of both latent variables – WOM and risk aversion – for each agent in a year. While applying Eq. (4), we calculate WOM followed by risk aversion to ensure the loading of WOM on risk aversion is explicitly accounted. By plugging in latent variable values, the purchase price, and city-level and social-network-level AV adoption in Eq. (8), we compute the systematic part of utility. Subsequently, the probability of choosing the AV by each agent in a year is obtained. The probability calculation involves the computation of the cumulative distribution function of normal distribution. The *probability* of adopting an AV is translated into the *decision to buy* an AV if the probability is greater than the random number drawn from a uniform distribution with a range (0.5, 1).

Once an agent purchases an AV, the decision to replace the AV is determined probabilistically based on the user-reported value in the survey. Specifically, if the duration of AV ownership exceeds the user-reported vehicle purchase frequency, the decision to replace the AV (or re-enter the AV market) is determined probabilistically based on a uniform random number (re-enter the market of random number is greater than 0.5). The density plot of the user-reported vehicle replacement time period is shown in Fig. 9. The plot indicates a high level of heterogeneity in vehicle purchase frequency of the individuals with a substantial mass concentrated in the region of 2-13 years.

Upon agents' purchases of AV, their post-purchase AV experience (positive/negative) is updated only once in the subsequent year after the purchase. Since there is no systematic way to capture post-purchase experiences of novel products like AVs, we use a random number generation approach. Specifically, we fix the proportion of agents with post-purchase satisfaction at the beginning of the simulation (see the variable *proportion of satisfied agents* in Table 7). We draw a random number from standard uniform distribution and assign positive experience to an agent if the random number is below the pre-specified proportion of satisfied agents, else negative experience is assigned. Depending upon the type of experience (positive or negative), an agent is assigned with a maximum or a minimum of her social network's WOM value. This WOM updating ensures that the type of post-purchase experience is fixed for an agent, but the WOM value (a proxy for the extent of positive/negative experience) may evolve over years.

Finally, social network- and city-level AV adoption levels are updated at the end of each year. The entire simulation is repeated for 50 iterations (i.e., 2021-2070) for any given scenario using fifteen different starting values (i.e., random seeds). We observed that the standard deviations of the forecasted market shares across different starting values were small, we only show the average of the city-level AV adoption across different starting values. Multiple simulation scenarios are considered by varying control variable values in Table 7.

#### 6.3. Agent-Based Simulation Results

We forecast the market share of AVs on a fifty-year horizon (2021-2070) in three scenarios characterized by a combination of i) AV



Fig. 8. Agent-Based Simulation framework using Interdependent ICLV model.



Fig. 9. Probability density function of vehicle purchase frequency.

price, ii) the extent of the post-purchase satisfaction of early adopters, iii) reduction in AV-involved crash rates, and clear liability in AV-involved crashes. All three scenarios are simulated for four different social network configurations of social networks, i.e., a combination of the two social distance measures (Euclidian/spatial and Gower) and two values of social ties (five and ten). The value of incorporating information dissemination in forecasting models is also highlighted.

## 6.3.1. Scenario 1: Effect of AV price and post-purchase satisfaction of adopters

We consider the starting price of AVs to be \$60,000 and assume it to decrease annually by 1%, 5%, and 10% (Bansal and Kockelman, 2017). Moreover, we forecast AV market share for the proportion of adopters with a positive post-purchase experience being either 30% or 90%. The plots of the forecasted AV adoption are shown in Fig. 10 (see Figures A.2 to A.5 in Supplementary sheet for plots with 95% confidence interval). Two main insights can be derived from these plots. First, there is a noticeable difference in the adoption trajectories obtained using spatial-distance- and Gower-similarity-based weight matrices. Second, the spatial-distance-based weight matrix configuration project a higher AV adoption rate than the Gower-similarity configuration across all price reduction rates and proportions of satisfied adopters. A possible explanation for such adoption trajectories lies in the fundamental nature of social network construction. In case of Gower-similarity weight matrix, an individual's social network contains ties which may disseminate very similar WOM (both magnitude and direction), leading to slower increase in overall WOM and slower decrease in risk aversion. These differences in trajectories are much more pronounced at lower price reduction rates (1%). The results indicate that price reduction of AV technology has a strong effect on AV adoption. Whereas a 5% annual reduction in AV price can help achieve a market share of around 75% in the next thirty years, an annual reduction of 1% would lead to an AV market share of only around 16% in 2050 (spatial-distance based social network with 5 ties and 30% satisfaction rate). Consumer satisfaction also plays a critical role in AV adoption. In case of moderate annual reduction of 5% in AV price and the same social network configuration, 50% market share is forecasted to be achieved in 16 years in a 90% post-purchase satisfaction scenario, but the same share would be attained in 18 years if 30% of early adopters are satisfied after buying AVs. The most effective strategy seems to be technological improvements to achieve large price reduction for a steep adoption trajectory.

### 6.3.2. Scenario 2: Effect of interpersonal social network

To capture the impact of the interpersonal social network, we benchmark the forecasting results of ABM by integrating interdependent and independent consumer behavior models. The independent model does not (explicitly) account for the interpersonal network effects (i.e.,  $\delta = 0$ ). Even in the absence of network effects, the consumer behavior model captures the direct impact of information sources on consumer uptake but fails to account for the indirect network effects transmitted through the structural equation of the WOM latent variable. The independent model cannot capture explicit WOM dissemination by (dis)satisfied adopters (i.e., the model is insensitive to the proportion of satisfied adopters), but can help capture the aggregate city and social-network adoption rate. We estimated a separate independent ICLV model by setting  $\delta = 0$ , and the estimation results are available in Tables A.8 to A.10 of the supplementary sheet. In Fig. 11 and Fig. 12, we present the adoption trajectories obtained based on interdependent ICLV models with Gower-similarity and spatial-distance based weight matrix configurations (with five ties), respectively. Similar trajectories based on the independent ICLV model are superimposed in these plots. Please note that the social network is also constructed for independent ICLV model to determine the city and social-network adoption percentages as they are used as explanatory variables in the choice model. These plots suggest that the adoption trajectory of independent ICLV model lies in between those of Gower-similarity (lower end) and spatial-distance (upper end) weight matrix configuration for a given price-rate reduction and proportion of satisfied adopters. The same pattern is also observed for the ten ties case (see Figures A.6 and A.7 in the supplementary sheet). Overall, the results (Fig. 11

## Table 7

Control	variables	and	their	descri	ntion
Control	variabics	anu	uncir	ucscii	puon

Variable	Description
Starting price	Starting market price of AVs
Yearly discount rate	Discount rate which determines the annual reduction
	in the AV price
Proportion of satisfied agents	Percentage of agents who are satisfied with the
	purchase of AV
Curvature crash	Curvature value used in the function to determine the rate
	of reduction in AV-involved crashes
Curvature legal	Curvature value used in the function to determine the rate
	of reduction in legal issues related to AV-involved crashes

and Fig. 12) suggest that ignoring explicit information dissemination may lead to under/over estimation of AV adoption depending on the underlying social-network configuration. These results corroborate our initial assertion that independent model may lead to unreliable forecasts due to its inability to capture the information propagation.

## 6.3.3. Scenario 3: Effect of reduction in AV-involved crashes and related legal issues

In this scenario, we quantify the effect of reduction in AV-involved crashes and the proportion of such crashes with no clear liability. As technology advances, AV-involved crashes and legal issues are likely to decrease, and the reduction is controlled in the ABM by curvature parameters (i.e., curvature crash and curvature legal) in Table 7. In particular, we consider four combinations of reductions in AV-involved crashes and legal issues regarding liability: (low, low), (low, high), (high, low), and (high, high) with corresponding curvatures (1.10, 5.00), (1.10, 1.00), (0.30, 5.00), and (0.30, 1.00), respectively. The AV-involved crashes and proportion of such crashes with legal issues in any year *t* can be obtained by plugging these values into the following function: Upperlimit –  $(\sqrt{\text{Upperlimit}}) * (t/\text{Curvature}) - 1$  (as discussed in section 5.2). In this scenario, we set an initial AV price to \$60,000, an annual reduction in AV price to 5%, and the proportion of satisfied adopters to 90%. Fig. 13 shows the impact of all four possible combinations of curvatures on the adoption of AVs (see Figures A.8 to A.11 in the supplementary sheet for plots with a 95% confidence interval). Similar to the previous observations, spatial-distance based weight matrix configuration provides a steeper adoption trajectory, *ceteris paribus*. The forecasting results show that the (high, high) scenario takes 25 years to achieve an AV market share of 50%, but the (low, low) scenario would attain the same market share in 27 years (based on ten ties Gower-similarity configuration).

#### 6.3.4. AV adoption in Nashville

We apply the ABM to assess the spatial (zip code-level) distribution of AV adoption levels in Nashville. We assume that AVs are introduced in 2021 with an initial price of \$60,000. We also consider an annual reduction in AV price by 5% and the proportion of satisfied adopters as 75%. The forecasted AV adoption densities (the number of adopted AVs per square mile) in the years 2030, 2040 and 2050 for Gower distance-based social network with five ties and with and without ( $\delta = 0$ ) explicit information dissemination is shown in Fig. 14 and Fig. 15. The plots indicate that, regardless of starting density, all zip codes exhibit marked improvement in AV density with time. Further, similar to the earlier observations, the independent ICLV model ( $\delta = 0$ ) has a higher rate of adoption as compared to the model with explicit information dissemination. In fact, the difference is very pronounced at the end of first 10 years (2030). The spatial adoption distribution for other models are provided in Figures A.12 to A.17 of the supplementary sheet .

#### 7. Conclusions and Future Work

Forecasting the adoption of the novel or "really new" products is of interest across multiple disciplines. Due to the lack of first-hand experience, potential early adopters of novel products actively expose themselves to word of mouth (WOM) information obtained from multiple sources (e.g., social networks and media) to assuage the potential risks. Not only do the existing consumer behavior models for novel products fail to capture the social network effects and interplay between WOM dissemination and risk aversion of consumers, but even cutting-edge forecasting models for such products also lack the integration of a calibrated consumer behavior model.

This study develops a general framework to forecast the adoption of novel products by combining a consumer behavior model with a population-level agent-based model (ABM). The consumer behavior is estimated using an integrated choice and latent variable (ICLV) model with WOM and risk aversion as two latent variables. The latent construct in ICLV captures spatial effects, i.e., the impact of homophilic peers in consumers' social networks. The discrete choice model in the ICLV accounts for the product's purchase price and adoption rates at individual's social network and city levels. We extend this ICLV model to account for the moderation of risk aversion behavior through WOM and panel effects in the choice component, and derive a maximum likelihood estimator to estimate the extended model. A calibrated consumer behavior model, synthetic population, and social network of all agents are embedded in an



Fig. 10. Effect of AV price reduction and proportion of AV adopters with post-purchase satisfaction on AV adoption.

85



Fig. 11. Effect of interpersonal social network on AV adoption for 5 ties configuration based on Gower-similarity weight matrix.



Fig. 12. Effect of interpersonal social network on AV adoption for 5 ties configuration based on spatial-distance weight matrix.

87



Fig. 13. Effect of reduction in AV-involved crashes and proportion of such crashes with legal issues on AV adoption.

agent-based simulation framework to forecast the market share of the novel product. The proposed framework is applied to forecast the future adoption of autonomous vehicles (AV) in Nashville, USA over the next thirty years.

We calibrate the consumer behavior model with the data collected from an online stated preference survey of 1,495 residents from Nashville. The consumer behavior model identifies the risk-averse demographic groups and those with a tendency to spread the positive WOM about AVs. The results also indicate that reduction in AV-involved crashes and clear liability in these crashes is critical for positive WOM dissemination. Positive WOM turns out to be the driving factor in reducing the risk aversion of consumers. The results of the discrete choice component show that lower purchase price and higher social-network- and city-level adoption increase an individual's likelihood to purchase an AV. Whereas results of the consumer behavior model are new additions to the AV literature, the main contribution of this study stems from providing a general framework to quantify the effect of policy interventions on the adoption of AV technology.

The calibrated consumer model and synthetic population of Nashville is passed through the ABM to forecast the market share of AVs over the next 50 years (i.e., 2021-2070) in different scenarios. We quantify the effect of reduction in the purchase price, the extent of the post-purchase satisfaction, the importance of including social network effects, and the improvement in AV safety measures on the market share of AVs. In a moderate scenario – an annual price reduction of 5% at the initial price of \$60,000, and 90% of adopters with post-purchase satisfaction, AVs are likely to attain 50% market share in eighteen years and around 80% market share in thirty-one years after their market introduction.

This study could be extended in several empirical and methodological directions. First, the agent-based simulation does not consider the used vehicle market, shared AVs, and comprehensive vehicle transaction mechanisms. The choice experiments also do not consider vehicle-specific attributes such as make, model, and body type. Future studies can conduct new experiments to capture these aspects or use revealed preference data, develop corresponding econometric models, and integrate them into the agent-based model. Second, this study assumes that CAVs are likely to reduce travel time and emissions. While several studies support this hypothesis, another strand of literature argues that ease of travel, increased empty miles, a modal shift from public transport, and land use changes could lead to an increase in travel time and emissions as CAV adoption increases (Annema, 2020; Fagnant and Kockelman, 2014, 2018; Fagnant et al., 2016; Massar et al., 2021; Wadud et al., 2016). Considering such negative implications of CAV adoption and accounting for their effect on CAV adoption trajectory are important avenues for future research. Third, we do not estimate the weight matrix in the ICLV estimation; instead, it is calculated based on the Euclidian distance or similarity in demographic characteristics. Some studies in spatial statistics have estimated weight matrix elements, but those are not scalable (Bhattacharjee and Jensen-Butler, 2013; Qu and Lee, 2015). Future studies can build upon these studies to find ways to reduce the complexity by assuming zero effect (i.e., fixing the weight matrix element to zero) beyond a certain distance or the number of social ties. Fourth, we consider a fixed number of ties for every individual in the ICLV estimation. Subsequent studies can explore potential ways to estimate the number of ties. The social networks in such subsequent studies can also consider familiarity or experience with AVs to bolster similarity between two consumers. Fifth, while we adopt CML due to its well-established statistical properties, future studies can explore the potential of alternative



Fig. 14. AV adoption density in Nashville, TN (a) 2030 (b) 2040 (c) 2050 (weight matrix based on Gower distance with five social ties with information dissemination).



Fig. 15. AV adoption density in Nashville, TN (a) 2030 (b) 2040 (c) 2050 (weight matrix based on Gower distance with five social ties with no information dissemination).

estimation methods such as the Bayesian Gibbs sampler. Since the model has Gaussian distributions at all levels, sampling from conditional posterior distributions of all model parameters should be straightforward, following the data augmentation techniques adopted by Daziano (2015) and Buddhavarapu et al. (2021). Moreover, emerging approximate Bayesian inference methods could be derived for the computationally efficient estimation of the proposed ICLV model. These methods are generally appropriate for initial specification search because they are around 20-50 times faster than MCMC (see Bansal et al. (2021a) for spatial count data models), but underestimate standard errors. Sixth, the simulation framework in our study relies on the synthetic population generated from the survey sample, which very well matches the current population of the study area. However, the temporal evolution of demographic variables is limited, leading to small standard errors of the AV adoption trajectories across repeated simulations. Future studies can complement the developed framework with the synthetic population for which several individuals- and household-level demographic characteristics could evolve using underlying econometric models (Eluru et al., 2008; Pinjari et al., 2008). Seventh, due to the unavailability of data on the impact of word of mouth and risk perceptions on the adoption of any other novel product, we could not validate the proposed framework. Future studies can validate the framework by collecting relevant data on social network effects and risk perceptions and applying it to forecast the adoption of a novel product with a market presence of a few years.

#### **CRediT** authorship contribution statement

Subodh Dubey: Conceptualization, Methodology, Data curation, Software, Formal analysis, Writing - original draft. Ishant

Sharma: Conceptualization, Methodology, Data curation, Software, Formal analysis, Writing – original draft. Sabyasachee Mishra: Writing – review & editing, Resources, Funding acquisition. Oded Cats: Writing – review & editing, Resources, Funding acquisition. Prateek Bansal: Conceptualization, Methodology, Writing – original draft, Supervision, Project administration.

## Acknowledgement

This research is partly funded by the Tennessee Department of Transportation (grant number RES-2018-06), Freight Mobility Research Institute, and the Center for Transportation Innovations in Education and Research (C-TIER) at the University of Memphis. The open-access for this research is supported by the CriticalMaaS project (grant number 804469), which is financed by the European Research Council and the Amsterdam Institute of Advanced Metropolitan Solutions.

The authors are grateful to associate editor and anonymous reviewers for their constructive feedback.

## Supplementary materials

Supplementary material associated with this article can be found, in the online version, at doi:10.1016/j.trb.2022.09.009.

## Appendix

```
Algorithm A.1 An algorithm to generate M_Diffmatrix.

\mathbf{M}_{\mathbf{D}} \mathbf{i} \mathbf{f} = \operatorname{zeros} \left( Q \left( H + T(I-1) \right), Q \left( H + TI \right) \right)
```

## for q in 1:Q

```
row ms = (q-1)*(H+T(I-1))+1
row me = q * (H + T(I - 1))
col_ms = (q-1)*(H+TI)+1
col me = q * (H + TI)
\mathbf{M}_{a} = \operatorname{zeros}(H + T(I-1), H + TI)
\mathbf{M}_{a}[1:H,1:H] = \mathbf{IDEN}_{H}
Iden mat = \mathbf{1}_{I-1}
O neg = -1*ones(I-1,1)
for t in 1:T
    if(i_a^t == 1)
       temp = O neg \sim Iden mat
    elseif(i_a^t == I)
       temp = Iden mat \sim O neg
    else
       temp = Iden_mat[.,1:i_a^t-1] ~ O_neg~Iden_mat[.,i_a^t:I-1]
    end
    row s = H + (t-1)*(I-I) + I
    row e = H + t^{*}(I - I)
    col s = H + (t-1)*I + I
    \operatorname{col} \mathbf{e} = H + \mathbf{t} * I
    \mathbf{M}_{a} [row s:row e,col s:col e] = temp
\mathbf{M}_{\mathbf{Diff}}[row_{ms}:row_{me}, col_{ms}:col_{me}] = \mathbf{M}_{a}
```

*Note*: where "~" refers to horizontal concatenation and  $i_a^t$  is the chosen alternative at time period t by individual q.

## A.2. Explicit form of CML function

The explicit form of CML function (Eq. 11) for a pair of individuals q and q' can be written with the help of additional matrices. In order to write the Eq. (11) in an expanded form, first use the algorithm decsribed above (A.1) to obtain the joint distribution in the utility differenced space (Eq. 10). Next, define following selection matrices.

Create a selection matrix  $\mathbf{D}_{qq'}$  of size  $[2(H + T(I - 1)) \times Q(H + T(I - 1))]$  to extract the mean  $(\overline{B}_{qq'})$  and covariance matrix  $(\overline{\Omega}_{qq'})$  for the pair of individuals q and q', such that  $\overline{B}_{qq'} = \mathbf{D}_{qq'} \overline{B}, \overline{\Omega}_{qq'} = \mathbf{D}_{qq'} \overline{\Omega} \mathbf{D}_{qq'}$ . To create  $\mathbf{D}_{qq'}$ , we create a matrix of zeros of the same size and insert two identity matrices of size (H + T(I - 1)). The first identify matrix is inserted in the first (H + T(I - 1)) rows and columns  $[(q - 1)^*(H + T(I - 1)) + 1]$  to  $[q^*(H + T(I - 1))]$  of  $\mathbf{D}_{qq'}$ . The second identity matrix is inserted in rows (H + T(I - 1) + 1) to [2(H + T(I - 1))], and columns  $[(q' - 1)^*(H + T(I - 1)) + 1]$  to  $[q^{(*}(H + T(I - 1))]$ .

Next, to explicitely write the *first term* (pairing of ordinal variables) of the CML expression in Eq. (11), we create a selection matrix  $V_{qq'}$  of size ( $2H \times 2(H + T(I - 1))$ ). Specifically, we create a matrix of zeros of the same size as of  $V_{qq'}$  and insert two identity matrices of size ( $H \times H$ ). The first identity matrix is inserted in first *H* rows and columns of  $V_{qq'}$  and insert another identity matrix in rows (H + 1) to 2*H* and columns [H + T(I - 1) + 1]to[2H + T(I - 1)].

$$\begin{split} \overline{B}_{\mathbf{V},qq'} &= \mathbf{V}_{qq'}\overline{B}_{qq'}, \overline{\Omega}_{\mathbf{V},qq'} = \mathbf{V}_{qq'}\overline{\Omega}_{qq'}\mathbf{V}_{qq'}, \\ \mathbf{\Psi}_{qq',low} &= (\mathbf{\Psi}_{low}[\{(q-1)*H+1\}:\{(q-1)*H+H\}], \mathbf{\Psi}_{low}[\{(q'-1)*H+1\}:\{(q'-1)*H+H\}])', \\ \mathbf{\Psi}_{qq',up} &= (\mathbf{\Psi}_{up}[\{(q-1)*H+1\}:\{(q-1)*H+H\}], \mathbf{\Psi}_{up}[\{(q'-1)*H+1\}:\{(q'-1)*H+H\}])', \\ \mathbf{\mu}_{\mathbf{V},low} &= \frac{\mathbf{\Psi}_{qq',low} - \overline{B}_{\mathbf{V},qq'}}{\sqrt{diag(\overline{\Omega}_{\mathbf{V},qq'})}}, \\ \mathbf{\mu}_{\mathbf{V},up} &= \frac{\mathbf{\Psi}_{qq',up} - \overline{B}_{\mathbf{V},qq'}}{\sqrt{diag(\overline{\Omega}_{\mathbf{V},qq'})}} \end{split}$$

To explicitely write the *second term* (pairing of ordinal variables with nominal variables) of the CML expression in Eq. (11), we create a rearrangement matrix  $\Delta_{qq'}$  of size  $[2(H + T(I - 1)) \times 2(H + T(I - 1))]$  that brings together ordinal and nominal responses together for simplicity. We create a matrix of zeros of the same size as of  $\Delta_{qq'}$  and insert four identity matrices, two of size  $(H \times H)$  and the remaining two of size  $[T(I - 1) \times T(I - 1)]$ . The first identity matrix of size  $(H \times H)$  is inserted in the first *H* rows and columns, and the second identity matrix in rows (H + 1) to 2H and columns [H + T(I - 1) + 1] to [2H + T(I - 1)] of  $\Delta_{qq'}$ . Similarly, the first identity matrix of size  $[T(I - 1) \times T(I - 1)]$  in inserted in rows (2H + 1) to [2H + T(I - 1)] and columns (H + 1) to [H + T(I - 1)] and the second identity matrix in rows [2H + T(I - 1) + 1] to [2(H + T(I - 1))] and columns [2H + T(I - 1) + 1] to [2(H + T(I - 1))] and the second identity matrix in rows [2H + T(I - 1) + 1] to [2(H + T(I - 1))] and columns [2H + T(I - 1) + 1] to [2(H + T(I - 1))] is also defined. To create  $H_{ht}$ , we start with a matrix of zeros of the same size. Subsequently, the cell (1, h) is filled with value of 1 and an identity matrix of size  $(I - 1) \times (I - 1)$  is inserted in rows 2 to I and columns [2H + (t - 1)(I - 1) + 1] through [2H + t(I - 1)]. Then we define the following notations:

$$\overline{\boldsymbol{B}}_{\boldsymbol{\Delta},qq'} = \boldsymbol{\Delta}_{qq'} \overline{\boldsymbol{B}}_{qq'}, \overline{\boldsymbol{\Omega}}_{\boldsymbol{\Delta},qq'} = \boldsymbol{\Delta}_{qq'} \overline{\boldsymbol{\Omega}}_{qq'} \boldsymbol{\Delta}_{qq'}, \\ \boldsymbol{\psi}_{low}^{h} = \left(\boldsymbol{\psi}_{qq',low}, \boldsymbol{0}_{T(I-1)}\right), \boldsymbol{\psi}_{up}^{h} = \left(\boldsymbol{\psi}_{qq',up}, \boldsymbol{0}_{T(I-1)}\right).$$

Finally, to write the *third term* (pairing of nominal variables) in Eq. 11 explicitly, we create a matrix  $\mathbf{E}_{tt}$  of size  $[2(I-1) \times 2(H+T(I-1))]$ . Specifically, we create a matrix of zeros of the same size as of  $\mathbf{E}_{tt}$  and insert two identity matrices of size  $[(I-1) \times (I-1)]$ . The first identity matrix is inserted in first (I-1) rows and columns [2H + (t-1)(I-1) + 1]through [2H + t(I-1)]. The second identity matrix is inserted in the last (I-1) rows and columns [2H + (t'-1)(I-1) + 1]through [2H + t'(I-1)]. The surrogate likelihood function a pair of individuals q and q' is:

$$L_{CML}^{q\bar{q}}(\Theta) = \left(\prod_{h=1}^{2H-1}\prod_{h'=h+1}^{2H}\left[\Phi_{2}\left\{\left\{\boldsymbol{\mu}_{\mathbf{V},up}[h],\boldsymbol{\mu}_{\mathbf{V},up}[h'']\right\},\overline{\boldsymbol{\Omega}}_{\mathbf{V},qq'}^{hh'}\right\} - \Phi_{2}\left(\left\{\boldsymbol{\mu}_{\mathbf{V},up}[h],\boldsymbol{\mu}_{\mathbf{V},low}[h'']\right\},\overline{\boldsymbol{\Omega}}_{\mathbf{V},qq'}^{hh'}\right)\right)\right) \times \left(\prod_{h=1}^{2H}\prod_{t=1}^{2T}\Phi_{I}\left[\left(\mathbf{H}_{ht}\boldsymbol{\psi}_{up}^{h} - \mathbf{H}_{ht}\overline{\boldsymbol{B}}_{\Delta,qq'}\right),\mathbf{H}_{ht}\overline{\boldsymbol{\Omega}}_{\Delta,qq'}\mathbf{H}_{ht}\right] - \Phi_{I}\left[\left(\mathbf{H}_{ht}\boldsymbol{\psi}_{low}^{h} - \mathbf{H}_{ht}\overline{\boldsymbol{B}}_{\Delta,qq'}\right),\mathbf{H}_{ht}\overline{\boldsymbol{\Omega}}_{\Delta,qq'}\mathbf{H}_{ht}\right]\right) \times \left(\prod_{t=1}^{2T-1}\prod_{t=t+1}^{2T}\Phi_{2(t-1)}\left[-\mathbf{E}_{tt}\overline{\boldsymbol{B}}_{qq'},\mathbf{E}_{tt}^{i}\overline{\boldsymbol{\Omega}}_{qq'}\mathbf{E}_{tt}^{i}\right]\right).$$
(A.1)

where  $\overline{\boldsymbol{\Omega}}_{\mathbf{V},q\dot{q}}^{hh}$  is a (2 × 2) submatrix of  $\overline{\boldsymbol{\Omega}}_{\mathbf{V},q\dot{q}}$  corresponding to *h* and *h*'ordinal indicators, and  $\Phi_{2(I-1)}[-\mathbf{E}_{tt}\cdot\overline{\boldsymbol{B}}_{q\dot{q}}\cdot\mathbf{E}_{tt'}]$  is a cummulative distribution function of 2(*I* - 1) dimensional MVN with mean  $(-\mathbf{E}_{tt}\cdot\overline{\boldsymbol{B}}_{q\dot{q}'})$  and covariance  $(\mathbf{E}_{tt}\cdot\overline{\boldsymbol{\Omega}}_{q\dot{q}}\cdot\mathbf{E}_{tt'})$ .

The maximum dimension of integration in Eq. (A.1) is (I - 1)\*2. We use GHK- simulator with quasi-random draws to evaluate MVNCDF accurately. Readers are referred to Train (2000) and Bhat (2003) for a detailed discussion on benefits of using quasi-random draws. The asymptotic covariance matrix of the model parameters is obtained by inverting (Godambe, 1960) sandwich information matrix, which requires Hessian and Jacobian matrices of the loglikelihood function at the convergence. The Jacobian matrix for models with spatial dependencies is computed using window sampling approach. Readers are referred to Zhao and Joe (2005), Bhat (2014), and Sidharthan and Bhat (2012) for more details on the calculations of the Hessian and the Jacobian matrix.

#### References

- Alzahrani, K., Hall-Phillips, A., Zeng, A.Z., 2019. Applying the theory of reasoned action to understanding consumers' intention to adopt hybrid electric vehicles in Saudi Arabia. Transportation (Amst) 46, 199–215. https://doi.org/10.1007/s11116-017-9801-3.
- Annema, J.A., 2020. Policy implications of the potential carbon dioxide (CO2) emission and energy impacts of highly automated vehicles. Advances in Transport Policy and Planning. Elsevier, pp. 149–162.

Anselin, L., 2013. Spatial econometrics: methods and models. Springer Science & Business Media.

Anselin, L, 2010. Thirty years of spatial econometrics. Papers in regional science 89, 3-25. https://doi.org/10.1111/j.1435-5957.2010.00279.x.

Baber, A., Thurasamy, R., Malik, M.I., Sadiq, B., Islam, S., Sajjad, M., 2016. Online word-of-mouth antecedents, attitude and intention-to-purchase electronic products in Pakistan. Telematics and Informatics 33, 388–400. https://doi.org/10.1016/j.tele.2015.09.004.

Bansal, P., Kockelman, K.M., 2018. Are we ready to embrace connected and self-driving vehicles? A case study of Texans. Transportation (Amst) 45, 641–675. https://doi.org/10.1007/s11116-016-9745-z.

Bansal, P., Kockelman, K.M., 2017. Forecasting Americans' long-term adoption of connected and autonomous vehicle technologies. Transp Res Part A Policy Pract 95, 49–63. https://doi.org/10.1016/j.tra.2016.10.013.

Bansal, P., Kockelman, K.M., Singh, A., 2016. Assessing public opinions of and interest in new vehicle technologies: An Austin perspective. Transp Res Part C Emerg Technol 67, 1–14. https://doi.org/10.1016/j.trc.2016.01.019.

Bansal, P., Krueger, R., Graham, D.J., 2021a. Fast Bayesian estimation of spatial count data models. Comput Stat Data Anal 157, 107152.

Bansal, P., Kumar, R.R., Raj, A., Dubey, S., Graham, D.J., 2021b. Willingness to pay and attitudinal preferences of Indian consumers for electric vehicles. Energy Econ 100, 105340. https://doi.org/10.1016/j.eneco.2021.105340.

Barham, B.L., Chavas, J.-P., Fitz, D., Salas, V.R., Schechter, L., 2014. The roles of risk and ambiguity in technology adoption. J Econ Behav Organ 97, 204–218. https://doi.org/10.1016/j.jebo.2013.06.014.

Baron, O., Berman, O., Nourinejad, M., 2021. Introducing Autonomous Vehicles: Adoption Patterns and Impacts on Social Welfare. https://doi.org/10.1287/ MSOM.2020.0955.

- Barth, M., Jugert, P., Fritsche, I., 2016. Still underdetected–Social norms and collective efficacy predict the acceptance of electric vehicles in Germany. Transp Res Part F Traffic Psychol Behav 37, 64–77. https://doi.org/10.1016/j.trf.2015.11.011.
- Bhandari, M., Rodgers, S., 2018. What does the brand say? Effects of brand feedback to negative eWOM on brand trust and purchase intentions. Int J Advert 37, 125–141. https://doi.org/10.1080/02650487.2017.1349030.
- Bhat, C.R., 2018. New matrix-based methods for the analytic evaluation of the multivariate cumulative normal distribution function. Transportation Research Part B: Methodological 109, 238–256. https://doi.org/10.1016/j.trb.2018.01.011.
- Bhat, C.R., 2014. The composite marginal likelihood (CML) inference approach with applications to discrete and mixed dependent variable models. University of Texas at Austin. Data-Supported Transportation Operations & Planning Center (D- STOP).
- Bhat, C.R., 2003. Simulation estimation of mixed discrete choice models using randomized and scrambled Halton sequences. Transportation Research Part B: Methodological 37, 837–855. https://doi.org/10.1016/S0191-2615(02)00090-5.
- Bhat, C.R., Astroza, S., Bhat, A.C., Nagel, K., 2016a. Incorporating a multiple discrete-continuous outcome in the generalized heterogeneous data model: Application to residential self-selection effects analysis in an activity time-use behavior model. Transportation Research Part B: Methodological 91, 52–76. https://doi.org/ 10.1016/j.trb.2016.03.007.
- Bhat, C.R., Dubey, S.K., Alam, M.J.Bin, Khushefati, W.H., 2015. A new spatial multiple discrete-continuous modeling approach to land use change analysis. J Reg Sci 55, 801–841. https://doi.org/10.1111/jors.12201.

Bhat, C.R., Pinjari, A.R., Dubey, S.K., Hamdi, A.S., 2016b. On accommodating spatial interactions in a generalized heterogeneous data model (GHDM) of mixed types of dependent variables. Transportation Research Part B: Methodological 94, 240–263.

- Bhattacharjee, A., Jensen-Butler, C., 2013. Estimation of the spatial weights matrix under structural constraints. Reg Sci Urban Econ 43, 617–634. https://doi.org/ 10.1016/j.regsciurbeco.2013.03.005.
- Bhattacharya, S., Sarkar, D., 2021. Study on Information Diffusion in Online Social Network. Proceedings of International Conference on Frontiers in Computing and Systems. Springer, pp. 279–288.
- Blanco, M., Atwood, J., Russell, S.M., Trimble, T., McClafferty, J.A., Perez, M.A., 2016. Automated vehicle crash rate comparison using naturalistic data. Virginia Tech Transportation Institute.
- Brick, K., Visser, M., 2015. Risk preferences, technology adoption and insurance uptake: A framed experiment. J Econ Behav Organ 118, 383–396. https://doi.org/ 10.1016/j.jebo.2015.02.010.
- Buddhavarapu, P., Bansal, P., Prozzi, J.A., 2021. A new spatial count data model with time-varying parameters. Transportation Research Part B: Methodological 150, 566–586. https://doi.org/10.1016/j.trb.2021.06.015.

Callegaro, M., Baker, R.P., Bethlehem, J., Göritz, A.S., Krosnick, J.A., Lavrakas, P.J., 2014. Online panel research: A data quality perspective. John Wiley & Sons. Chatura, R., 2015. The impact of word of mouth on service purchase decisions: Examining risk and the interaction of tie strength and involvement. Journal of Service

Theory and Practice 25, 636–656. https://doi.org/10.1108/JSTP-04-2014-0070.

Chen, J., Teng, L., Yu, Y., Yu, X., 2016. The effect of online information sources on purchase intentions between consumers with high and low susceptibility to informational influence. J Bus Res 69, 467–475. https://doi.org/10.1016/j.jbusres.2015.05.003.

Chen, M.F., 2007. Consumer attitudes and purchase intentions in relation to organic foods in Taiwan: Moderating effects of food-related personality traits. Food Qual Prefer 18, 1008–1021. https://doi.org/10.1016/j.foodqual.2007.04.004.

Cheung, C.M.K., Lee, M.K.O., Thadani, D.R., 2009. The impact of positive electronic word-of-mouth on consumer online purchasing decision. World Summit on Knowledge Society. Springer, pp. 501–510.

- Chikaraishi, M., Khan, D., Yasuda, B., Fujiwara, A., 2020. Risk perception and social acceptability of autonomous vehicles: A case study in Hiroshima. Japan. Transp Policy (Oxf) 98, 105–115. https://doi.org/10.1016/j.tranpol.2020.05.014.
- David-Barrett, T., 2020. Herding Friends in Similarity-Based Architecture of Social Networks. Scientific Reports 2020 10, 1–6. https://doi.org/10.1038/s41598-020-61330-6, 1 10.

Daziano, R.A., 2015. Inference on mode preferences, vehicle purchases, and the energy paradox using a Bayesian structural choice model. Transportation Research Part B: Methodological 76, 1–26. https://doi.org/10.1016/j.trb.2015.02.012.

Dholakia, U.M., 2001. A motivational process model of product involvement and consumer risk perception. Eur J Mark. https://doi.org/10.1108/ EUM0000000006479.

Din, A., Wilson, R., 2020. Crosswalking ZIP Codes to Census Geographies. Cityscape 22, 293-314.

Dowling, G.R., Staelin, R., 1994. A model of perceived risk and intended risk-handling activity. Journal of consumer research 21, 119–134. https://doi.org/10.1086/209386.

Du, H., Liu, D., Sovacool, B.K., Wang, Y., Ma, S., Li, R.Y.M., 2018. Who buys New Energy Vehicles in China? Assessing social-psychological predictors of purchasing awareness, intention, and policy. Transp Res Part F Traffic Psychol Behav 58, 56–69. https://doi.org/10.1016/j.trf.2018.05.008.

Eluru, N., Pinjari, A.R., Guo, J.Y., Sener, I.N., Srinivasan, S., Copperman, R.B., Bhat, C.R., 2008. Population updating system structures and models embedded in the comprehensive econometric microsimulator for urban systems. Transp Res Rec 2076, 171–182. https://doi.org/10.3141/2076-19.

Fagnant, D., Kockelman, K.M., 2014. The Travel and Environmental Implications of Shared Autonomous Vehicles, Using Agent-based Model Scenarios. Transportation Research Part C 40, 1–13. https://doi.org/10.1016/j.trc.2013.12.001.

Fagnant, D.J., Kockelman, K.M., 2018. Dynamic ride-sharing and fleet sizing for a system of shared autonomous vehicles in Austin, Texas. Transportation (Amst) 45, 143–158. https://doi.org/10.1007/s11116-016-9729-z.

Fagnant, D.J., Kockelman, K.M., Bansal, P., 2016. Operations of Shared Autonomous Vehicle Fleet for Austin, Texas, Market. Transp Res Rec 2563, 98–106. https:// doi.org/10.3141/2536-12.

Ghasri, M., Vij, A., 2021. The potential impact of media commentary and social influence on consumer preferences for driverless cars. Transp Res Part C Emerg Technol 127, 103132. https://doi.org/10.1016/j.trc.2021.103132.

Godambe, V.P., 1960. An optimum property of regular maximum likelihood estimation. The Annals of Mathematical Statistics 31, 1208–1211. https://doi.org/ 10.1214/aoms/1177705693.

Golbabaei, F., Yigitcanlar, T., Paz, A., Bunker, J., 2020. Individual Predictors of Autonomous Vehicle Public Acceptance and Intention to Use: A Systematic Review of the Literature. Journal of Open Innovation: Technology, Market, and Complexity 6, 106. https://doi.org/10.3390/joitmc6040106.

Gower, J.C., 1971. A General Coefficient of Similarity and Some of Its Properties. Biometrics 27, 857. https://doi.org/10.2307/2528823.

Granovetter, M., 1978. Threshold models of collective behavior. American journal of sociology 83, 1420–1443.

Gregan-Paxton, J., John, D.R., 1997. Consumer learning by analogy: A model of internal knowledge transfer. Journal of Consumer Research 24, 266–284. https://doi.org/10.1086/209509.

- Greszki, R., Meyer, M., Schoen, H., 2015. Exploring the effects of removing "too fast" responses and respondents from web surveys. Public Opin Q 79, 471–503. https://doi.org/10.1093/poq/nfu058.
- Grinblatt, M., Keloharju, M., Ikäheimo, S., 2008. Social influence and consumption: Evidence from the automobile purchases of neighbors. Rev Econ Stat 90, 735–753. https://doi.org/10.1162/rest.90.4.735.

Gupta, P., Harris, J., 2010. How e-WOM recommendations influence product consideration and quality of choice: A motivation to process information perspective. J Bus Res 63, 1041–1049. https://doi.org/10.1016/j.jbusres.2009.01.015.

Ha, H.-Y., 2002. The Effects of Consumer Risk Perception on Pre-purchase Information in Online Auctions: Brand, Word-of-Mouth, and Customized Information. Journal of Computer-Mediated Communication 8. https://doi.org/10.1111/j.1083-6101.2002.tb00160.x.

Haboucha, C.J., Ishaq, R., Shiftan, Y., 2017. User preferences regarding autonomous vehicles. Transp Res Part C Emerg Technol 78, 37–49. https://doi.org/10.1016/j. trc.2017.01.010.

Han, H., Hsu, L.T.J., Lee, J.S., Sheu, C., 2011. Are lodging customers ready to go green? An examination of attitudes, demographics, and eco-friendly intentions. Int J Hosp Manag 30, 345–355. https://doi.org/10.1016/J.IJHM.2010.07.008.

He, L., Wang, M., Chen, W., Conzelmann, G., 2014. Incorporating social impact on new product adoption in choice modeling: A case study in green vehicles. Transp Res D Transp Environ 32, 421–434. https://doi.org/10.1016/j.trd.2014.08.007.

Helveston, J.P., Liu, Y., Feit, E.M., Fuchs, E., Klampfl, E., Michalek, J.J., 2015. Will subsidies drive electric vehicle adoption? Measuring consumer preferences in the US and China. Transp Res Part A Policy Pract 73, 96–112. https://doi.org/10.1016/j.tra.2015.01.002.

Hijmans, R.J., Williams, E., Vennes, C., Hijmans, M.R.J., 2017. Package 'geosphere.' Spherical trigonometry 1.

Hilgarter, K., Granig, P., 2020. Public perception of autonomous vehicles: A qualitative study based on interviews after riding an autonomous shuttle. Transp Res Part F Traffic Psychol Behav 72, 226–243. https://doi.org/10.1016/J.TRF.2020.05.012.

Hirunyawipada, T., Paswan, A.K., 2006. Consumer innovativeness and perceived risk: implications for high technology product adoption. Journal of consumer marketing. https://doi.org/10.1108/07363760610674310.

Hohenberger, C., Spörrle, M., Welpe, I.M., 2016. How and why do men and women differ in their willingness to use automated cars? The influence of emotions across different age groups. Transp Res Part A Policy Pract 94, 374–385. https://doi.org/10.1016/j.tra.2016.09.022.

Hong, Z., Li, M., Han, X., He, X., 2020. Innovative green product diffusion through word of mouth. Transp Res E Logist Transp Rev 134, 101833. https://doi.org/ 10.1016/j.tre.2019.101833.

Hsu, C.I., Li, H.C., Lu, S.M., 2013. A dynamic marketing model for hybrid electric vehicles: A case study of Taiwan. Transp Res D Transp Environ 20, 21–29. https://doi.org/10.1016/j.trd.2013.01.001.

Huang, L., Zhang, J., Liu, H., Liang, L., 2014. The effect of online and offline word-of-mouth on new product diffusion. Journal of Strategic Marketing 22, 177–189. https://doi.org/10.1080/0965254X.2013.876070.

Huete-Alcocer, N., 2017. A Literature Review of Word of Mouth and Electronic Word of Mouth: Implications for Consumer Behavior. Frontiers in Psychology. https://doi.org/10.3389/fpsyg.2017.01256.

Hussain, S., Ahmed, W., Jafar, R.M.S., Rabnawaz, A., Jianzhou, Y., 2017. eWOM source credibility, perceived risk and food product customer's information adoption. Comput Human Behav 66, 96–102. https://doi.org/10.1016/j.chb.2016.09.034.

Hussain, S., Guangju, W., Jafar, R.M.S., Ilyas, Z., Mustafa, G., Jianzhou, Y., 2018. Consumers' online information adoption behavior: Motives and antecedents of electronic word of mouth communications. Comput Human Behav 80, 22–32. https://doi.org/10.1016/j.chb.2017.09.019.

IIHS, 2020. Fatality Facts 2018 [WWW Document]. URL. https://www.iihs.org/topics/fatality-statistics/detail/state-by-state (accessed 6.23.20).

Iyer, R., Griffin, M., 2020. Modeling word-of-mouth usage: A replication. J Bus Res. https://doi.org/10.1016/j.jbusres.2019.12.027.

Jalilvand, M.R., Samiei, N., 2012. The effect of electronic word of mouth on brand image and purchase intention. Marketing Intelligence & Planning. https://doi.org/ 10.1108/02634501211231946.

- Jansson, J., Marell, A., Nordlund, A., 2011. Exploring consumer adoption of a high involvement eco-innovation using value-belief-norm theory. Journal of Consumer Behaviour 10, 51–60. https://doi.org/10.1002/CB.346.
- Jansson, J., Nordlund, A., Westin, K., 2017. Examining drivers of sustainable consumption: The influence of norms and opinion leadership on electric vehicle adoption in Sweden. J Clean Prod 154, 176–187. https://doi.org/10.1016/j.jclepro.2017.03.186.

Jerit, J., Barabas, J., Bolsen, T., 2006. Citizens, knowledge, and the information environment. Am J Pol Sci 50, 266–282. https://doi.org/10.1111/J.1540-5907.2006.00183.X.

Jung, N.Y., Seock, Y.-K., 2017. Effect of service recovery on customers' perceived justice, satisfaction, and word-of-mouth intentions on online shopping websites. Journal of Retailing and Consumer Services 37, 23–30. https://doi.org/10.1016/j.jretconser.2017.01.012.

Jung, S., 2015. Does education affect risk aversion? Evidence from the British education reform. Appl Econ 47, 2924–2938. https://doi.org/10.1080/00036846.2015.1011313.

Kaneko, M., Kagawa, S., 2021. Driving propensity and vehicle lifetime mileage: A quantile regression approach. J Environ Manage 278, 111499. https://doi.org/ 10.1016/J.JENVMAN.2020.111499.

Keane, M.P., 1992. A note on identification in the multinomial probit model. Journal of Business & Economic Statistics 10, 193–200. https://doi.org/10.2307/1391677.

Kieckhäfer, K., Wachter, K., Spengler, T.S., 2017. Analyzing manufacturers' impact on green products' market diffusion – the case of electric vehicles. J Clean Prod 162, S11–S25. https://doi.org/10.1016/j.jclepro.2016.05.021.

Kim, J., Rasouli, S., Timmermans, H., 2014. Expanding scope of hybrid choice models allowing for mixture of social influences and latent attitudes: Application to intended purchase of electric cars. Transp Res Part A Policy Pract 69, 71–85. https://doi.org/10.1016/j.tra.2014.08.016.

Kim, S.H., Circella, G., Mokhtarian, P.L., 2019. Identifying latent mode-use propensity segments in an all-AV era. Transp Res Part A Policy Pract 130, 192–207. https://doi.org/10.1016/J.TRA.2019.09.015.

Knight, J., Weir, S., Woldehanna, T., 2010. The role of education in facilitating risk-taking and innovation in agriculture 39, 1–22. https://doi.org/10.1080/ 00220380312331293567.

Konduri, K.C., You, D., Garikapati, V.M., Pendyala, R.M., 2016. Enhanced synthetic population generator that accommodates control variables at multiple geographic resolutions. Transp Res Rec 2563, 40–50. https://doi.org/10.3141/2563-08.

Krishnamurthy, S., 2001. A comprehensive analysis of permission marketing. Journal of Computer-Mediated Communication 6. https://doi.org/10.1111/j.1083-6101.2001.tb00119.x.

Kwon, Y., Son, S., Jang, K., 2020. User satisfaction with battery electric vehicles in South Korea. Transp Res D Transp Environ 82, 102306. https://doi.org/10.1016/j. trd.2020.102306.

- Kyriakidis, M., Happee, R., de Winter, J.C.F., 2015. Public opinion on automated driving: Results of an international questionnaire among 5000 respondents. Transp Res Part F Traffic Psychol Behav 32, 127–140. https://doi.org/10.1016/j.trf.2015.04.014.
- Lavieri, P.S., Garikapati, V.M., Bhat, C.R., Pendyala, R.M., Astroza, S., Dias, F.F., 2017. Modeling Individual Preferences for Ownership and Sharing of Autonomous Vehicle Technologies. Transportation Research Record: Journal of the Transportation Research Board 2665, 1–10. https://doi.org/10.3141/2665-01.
- Liljamo, T., Liimatainen, H., Pöllänen, M., 2018. Attitudes and concerns on automated vehicles. Transp Res Part F Traffic Psychol Behav 59, 24–44. https://doi.org/ 10.1016/j.trf.2018.08.010.
- Ma, S.-C., Fan, Y., Guo, J.-F., Xu, J.-H., Zhu, J., 2019. Analysing online behaviour to determine Chinese consumers' preferences for electric vehicles. J Clean Prod 229, 244–255. https://doi.org/10.1016/j.jclepro.2019.04.374.
- Mahajan, V., Muller, E., Kerin, R.A., 1984. Introduction strategy for new products with positive and negative word-of-mouth. Manage Sci 30, 1389–1404.
- Maness, M., Cirillo, C., Dugundji, E.R., 2015. Generalized behavioral framework for choice models of social influence: Behavioral and data concerns in travel behavior. J Transp Geogr 46, 137–150. https://doi.org/10.1016/j.jtrangeo.2015.06.005.
- Manning, K.C., Bearden, W.O., Madden, T.J., 1995. Consumer Innovativeness and the Adoption Process. Journal of Consumer Psychology 4, 329-345. https://doi.org/10.1207/s15327663jcp0404\_02.

Manson, S., Schroeder, J., Van Riper, D., Ruggles, S., 2019. IPUMS National Historical Geographic Information System: Version 14.0 [Database]. IPUMS, Minneapolis, MN, 2019.

- Marchand, A., Hennig-Thurau, T., Wiertz, C., 2017. Not all digital word of mouth is created equal: Understanding the respective impact of consumer reviews and microblogs on new product success. International Journal of Research in Marketing 34, 336–354. https://doi.org/10.1016/j.ijusemar.2016.09.003.
  Martin, W.C., Lueg, J.E., 2013. Modeling word-of-mouth usage. J Bus Res 66, 801–808. https://doi.org/10.1016/j.jbusres.2011.06.004.
- Martin, W.C., Eueg, J.E., 2013. Modeling word-on-motin usage. J Bus Kes 60, 801–806. https://doi.org/10.1010/j.j.bustes.2011.00.004. Massar, M., Reza, I., Rahman, S.M., Abdullah, S.M.H., Jamal, A., Al-Ismail, F.S., 2021. Impacts of Autonomous Vehicles on Greenhouse Gas Emissions—Positive or Negative? Int J Environ Res Public Health 18, 5567. https://doi.org/10.3390/ijerph18115567.
- Mau, P., Eyzaguirre, J., Jaccard, M., Collins-Dodd, C., Tiedemann, K., 2008. The 'neighbor effect': Simulating dynamics in consumer preferences for new vehicle technologies. Ecological Economics 68, 504–516. https://doi.org/10.1016/j.ecolecon.2008.05.007.
- Menon, N., Barbour, N., Zhang, Y., Pinjari, A.R., Mannering, F., 2019. Shared autonomous vehicles and their potential impacts on household vehicle ownership: An exploratory empirical assessment. Int J Sustain Transp 13, 111–122. https://doi.org/10.1080/15568318.2018.1443178.
- Moons, I., De Pelsmacker, P., 2012. Emotions as determinants of electric car usage intention. Journal of Marketing Management 28, 195–237. https://doi.org/ 10.1080/0267257X.2012.659007.
- Musti, S., Kockelman, K.M., 2011. Evolution of the household vehicle fleet: Anticipating fleet composition, PHEV adoption and GHG emissions in Austin, Texas. Transp Res Part A Policy Pract 45, 707–720. https://doi.org/10.1016/j.tra.2011.04.011.
- Narayan, V., Rao, V.R., Saunders, C., 2011. How peer influence affects attribute preferences: A Bayesian updating mechanism. Marketing Science 30, 368–384. https://doi.org/10.1287/mksc.1100.0618.
- Olia, A., Abdelgawad, H., Abdulhai, B., Razavi, S.N., 2016. Assessing the Potential Impacts of Connected Vehicles: Mobility, Environmental, and Safety Perspectives. Journal of Intelligent Transportation Systems: Technology, Planning, and Operations 20, 229–243. https://doi.org/10.1080/15472450.2015.1062728.
- Ozaki, R., Sevastyanova, K., 2011. Going hybrid: An analysis of consumer purchase motivations. Energy Policy 39, 2217–2227. https://doi.org/10.1016/j. enpol.2010.04.024.
- Paridon, T.J., Carraher, S., Carraher, S.C., 2006. The income effect in personal shopping value, consumer self-confidence, and information sharing (word of mouth communication) research. Academy of Marketing Studies Journal 10, 107.
- Park, Y., Chen, J. V., 2007. Acceptance and adoption of the innovative use of smartphone. Industrial Management and Data Systems 107, 1349–1365. https://doi.org/ 10.1108/02635570710834009.
- Payre, W., Cestac, J., Delhomme, P., 2014. Intention to use a fully automated car: Attitudes and a priori acceptability. Transp Res Part F Traffic Psychol Behav 27, 252–263. https://doi.org/10.1016/j.trf.2014.04.009.
- Pew Research Center, 2022. Al and Human Enhancement: Americans' openness is tempered by a range of concerns.
- Pinjari, A., Eluru, N., Srinivasan, S., Guo, J.Y., Copperman, R., Sener, I.N., Bhat, C.R., 2008. Cemdap: Modeling and microsimulation frameworks, software development, and verification. Proceedings of the Transportation Research Board 87th Annual Meeting, Washington DC.
- Prendergast, G., Ko, D., Siu Yin, V.Y., 2010. Online word of mouth and consumer purchase intentions. Int J Advert 29, 687–708. https://doi.org/10.2501/ S0265048710201427.
- Qu, X., Lee, L., 2015. Estimating a spatial autoregressive model with an endogenous spatial weight matrix. J Econom 184, 209–232. https://doi.org/10.1016/j. ieconom.2014.08.008.
- Rasouli, S., Timmermans, H., 2016. Influence of Social Networks on Latent Choice of Electric Cars: A Mixed Logit Specification Using Experimental Design Data. Netw Spat Econ 16, 99–130. https://doi.org/10.1007/s11067-013-9194-6.
- Richards, T.J., Hamilton, S.F., Allender, W.J., 2014. Social networks and new product choice. Am J Agric Econ 96, 489–516. https://doi.org/10.1093/ajae/aat116. Roßmann, J., 2010. Data quality in web surveys of the German longitudinal election study 2009. 3rd ECPR Graduate Conference.
- Saeed, T.U., Burris, M.W., Labi, S., Sinha, K.C., 2020. An empirical discourse on forecasting the use of autonomous vehicles using consumers' preferences. Technol Forecast Soc Change 158, 120130. https://doi.org/10.1016/j.techfore.2020.120130.
- Schoettle, B., Sivak, M., 2014. A survey of public opinion about connected vehicles in the U.S., the U.K., and Australia. https://doi.org/10.1109/ ICCVE.2014.7297637.
- Sharma, I., Mishra, S., 2022a. Quantifying the consumer's dependence on different information sources on acceptance of autonomous vehicles. Transp Res Part A Policy Pract 160, 179–203. https://doi.org/10.1016/j.tra.2022.04.009.
- Sharma, I., Mishra, S., 2022b. Ranking preferences towards adopting autonomous vehicles based on peer inputs and advertisements. Transportation (Amst) 1–54. https://doi.org/10.1007/s11116-022-10304-w.
- Sharma, I., Mishra, S., 2020. Modeling consumers' likelihood to adopt autonomous vehicles based on their peer network. Transp Res D Transp Environ 87, 102509. https://doi.org/10.1016/j.trd.2020.102509.
- Sidharthan, R., Bhat, C.R., 2012. Incorporating Spatial Dynamics and Temporal Dependency in Land Use Change Models. Geogr Anal 44, 321–349. https://doi.org/ 10.1111/J.1538-4632.2012.00854.X.
- Simpson, J.R., Sharma, I., Mishra, S., 2022. Modeling trucking industry perspective on the adoption of connected and autonomous trucks. Research in Transportation Business & Management 100883. https://doi.org/10.1016/j.rtbm.2022.100883.
- Sinha, A., Chand, S., Wijayaratna, K.P., Virdi, N., Dixit, V., 2020. Comprehensive safety assessment in mixed fleets with connected and automated vehicles: A crash severity and rate evaluation of conventional vehicles. Accid Anal Prev 142, 105567. https://doi.org/10.1016/J.AAP.2020.105567.
- Talebian, A., Mishra, S., 2018. Predicting the adoption of connected autonomous vehicles: A new approach based on the theory of diffusion of innovations. Transp Res Part C Emerg Technol 95, 363–380. https://doi.org/10.1016/j.trc.2018.06.005.
- Talebian, A., Mishra, S., 2022. Unfolding the state of the adoption of connected autonomous trucks by the commercial fleet owner industry. Transportation Research Part E: Logistics and Transportation Review 158, 102616. https://doi.org/10.1016/j.tre.2022.102616.
- Tan, S.J., 1999. Strategies for reducing consumers' risk aversion in Internet shopping. Journal of consumer marketing. https://doi.org/10.1108/ 07363769910260515.
- Thøgersen, J., Ebsen, J.V, 2019. Perceptual and motivational reasons for the low adoption of electric cars in Denmark. Transp Res Part F Traffic Psychol Behav 65, 89–106. https://doi.org/10.1016/j.trf.2019.07.017.
- Tobler, W.R., 1970. A computer movie simulating urban growth in the Detroit region. Econ Geogr 46, 234–240. https://doi.org/10.2307/143141.
- Torlak, O., Ozkara, B.Y., Tiltay, M.A., Cengiz, H., Dulger, M.F., 2014. The effect of electronic word of mouth on brand image and purchase intention: An application concerning cell phone brands for youth consumers in Turkey. Journal of Marketing Development and Competitiveness 8, 61.

Train, K., 2000. Halton sequences for mixed logit.

- Tsiotsou, R., Alexandris, K., 2009. Delineating the outcomes of sponsorship: sponsor image, word of mouth, and purchase intentions. International Journal of retail & distribution Management 37, 358–369. https://doi.org/10.1108/09590550910948583.
- Varin, C., Vidoni, P., 2005. A note on composite likelihood inference and model selection. Biometrika 92, 519-528.
- Wadud, Z., Chintakayala, P.K., 2021. To own or not to own That is the question: The value of owning a (fully automated) vehicle. Transp Res Part C Emerg Technol 123, 102978. https://doi.org/10.1016/J.TRC.2021.102978.
- Wadud, Z., MacKenzie, D., Leiby, P., 2016. Help or hindrance? The travel, energy and carbon impacts of highly automated vehicles. Transp Res Part A Policy Pract 86, 1–18. https://doi.org/10.1016/j.tra.2015.12.001.
- Wang, S., Zhao, J., 2019. Risk preference and adoption of autonomous vehicles. Transp Res Part A Policy Pract 126, 215–229. https://doi.org/10.1016/j. tra.2019.06.007.
- Waymo, 2020. Our Journey- Waymo [WWW Document]. URL. https://waymo.com/journey/ (accessed 11.19.20).
- Zhang, T., Gensler, S., Garcia, R., 2011. A study of the diffusion of alternative fuel vehicles: An agent-based modeling approach. Journal of Product Innovation Management 28, 152–168. https://doi.org/10.1111/j.1540-5885.2011.00789.x.
- Zhang, T., Tao, D., Qu, X., Zhang, X., Lin, R., Zhang, W., 2019. The roles of initial trust and perceived risk in public's acceptance of automated vehicles. Transp Res Part C Emerg Technol. https://doi.org/10.1016/j.trc.2018.11.018.
- Zhao, Y., Joe, H., 2005. Composite likelihood estimation in multivariate data analysis. Canadian Journal of Statistics 33, 335–356. https://doi.org/10.1002/ cjs.5540330303.
- Zoellick, J.C., Kuhlmey, A., Schenk, L., Schindel, D., Blüher, S., 2019. Amused, accepted, and used? Attitudes and emotions towards automated vehicles, their relationships, and predictive value for usage intention. Transp Res Part F Traffic Psychol Behav 65, 68–78. https://doi.org/10.1016/J.TRF.2019.07.009.